Department of **Social Policy** 

LSE

Working Paper July 2023

# **Perceptions Matter:**

Quasi-Experimental Evidence on the Effects of Spain's New Minimum Income on Households' Financial Wellbeing

Eugenia Bilbao-Goyoaga

Department of Social Policy London School of Economics and Political Science Houghton Street London WC2A 2AE

Email: <a href="mailto:soc.pol.webteam@lse.ac.uk">soc.pol.webteam@lse.ac.uk</a>

Telephone: +44 (0)20 7955 6001

# lse.ac.uk/social-policy



Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

**To cite this paper**: Bilbao-Goyoaga, E. (2023) Perceptions Matter: Quasi-Experimental Evidence on the Effects of Spain's New Minimum Income on Households' Financial Wellbeing, Social Policy Working Paper 02-23, London: LSE Department of Social Policy.

# Perceptions Matter: Quasi-Experimental Evidence on the Effects of Spain's New Minimum Income on Households' Financial Wellbeing

Eugenia Bilbao-Goyoaga

London School of Economics and Political Science e.m.bilbao-goyoaga-zabala@lse.ac.uk

# Abstract

This paper examines Spain's minimum income scheme (MIS) introduced in 2020 and its impact on households' objective and subjective financial wellbeing. The study addresses two key motivations. First, there is no consensus on the effectiveness of MISs in improving households' financial wellbeing despite the renewed interest prompted by the Covid-19 pandemic, ecological transition and cost-of-living crisis. Second, existing literature primarily focuses on objective measures of financial wellbeing like monetary poverty, but it is crucial to understand how income improvements through MISs are perceived by households. These subjective perceptions play a significant role in people's health, productivity and decision-making and provide insights into adaptation mechanisms and spillover effects on non-recipients. The study uses Eurostat survey data aggregated at the national level from 2010 to 2022, employing a Synthetic Control Method analysis. Results show that during the initial year and a half of implementation, the policy had no statistically significant effect on households' material conditions (e.g. poverty rate, poverty gap and mean income). However, after two and a half years, it did considerably improve how households perceive the evolution of their finances. The paper discusses mechanisms explaining this differential impact, including the policy's phased implementation, benefit enhancements from 2022 onwards as well as anticipation, placebo and positive spillover effects of the MIS. The findings highlight the importance for practitioners to consider subjective financial wellbeing when assessing MISs.

# Key words

Financial Wellbeing; Subjective Wellbeing; Minimum Income; Poverty; Quasi-Experiment; Synthetic Control Method

# **1** INTRODUCTION

This paper examines how Spain's new minimum income scheme affects households' financial wellbeing and whether this effect differs across objective material conditions and households' subjective perceptions. Minimum income schemes (MISs) are last-resort cash payments available to both working and non-working households who have insufficient financial means and are not entitled to contributory social insurance. These schemes are becoming increasingly relevant in the context of the Covid-19 pandemic, the ecological transition and the cost-of-living crisis as they can be used to support people who are either temporarily unable to work, who need time to learn new skills or whose wages are insufficient to cover living expenses (Cantillon *et al.*, 2019). Today, most countries have some form of MIS in place, although schemes differ in terms of how well they support beneficiaries. Countries like Denmark, Ireland or Slovakia have comprehensive MISs covering most households in need while others like Romania or Spain have weaker schemes with insufficient coverage (Almeida, Poli and Hernández, 2022). As a result, the Council of the European Union adopted in January 2023 a recommendation on common standards for adequate minimum incomes and now some practitioners are calling for a binding directive (Council of the European Union, 2023).

Despite the widespread use and relevance of MISs, a consensus does not exist among researchers and policymakers on the effectiveness of these schemes in improving the financial situation of households. In theory, giving cash to households should support them financially. However, the extent to which a MIS helps households depends on its various design elements, namely the generosity level and the coverage of people in need given its eligibility criteria, as authors such as Figari, Matsaganis and Sutherland (2013) or Almeida, Poli and Hernández (2022) have explained. The extent to which a MIS supports households also depends on behavioural responses such as the take-up of the benefit (e.g. Kleven and Kopczuk, 2011; Frazer and Marlier, 2016 in the EU) or the labour supply reactions of beneficiaries who might be disincentivised to work (e.g. Lemieux and Milligan, 2008 in Quebec; Bargain and Doorley, 2011 in France; Moffitt, 2016 in the US). Therefore, due to the complexity of predicting the overall impact of a particular MIS based on theory alone, this paper aims to conduct an impact evaluation of the newly implemented Spanish MIS.

However, empirically settling the debate on the effects of minimum income schemes (MISs) has proven challenging due to the non-random allocation of these schemes, which results in a lack of valid counterfactuals. The individuals benefiting from MISs and the countries

implementing comprehensive MISs possess certain characteristics that inherently impact financial wellbeing. This self-selection into the policy creates difficulties in isolating the true effects of a MIS. Academics have attempted to overcome this issue by employing ex-ante simulation models to analyse the impact of MISs on households' finances. However, these models rely on strong assumptions about the behavioural responses of beneficiaries. To address this gap, I conduct an ex-post analysis using a causal inference method, specifically the synthetic control approach. This approach allows me to avoid relying on behavioural assumptions and solves for self-selection issues.

When evaluating MISs, researchers as well as national and international policymakers focus on so-called 'objective measures of financial wellbeing'. They look at how income support affects households' income and, more specifically, poverty levels since these schemes are targeted at the lower end of the income distribution (Fleche et al., 2012; Xiao, 2013; Cantillon, Goedemé and Hills, 2019). Yet, while these objective measures are indeed central to households' financial wellbeing, they do not provide a complete picture of their experiences. 'Subjective financial wellbeing', which measures how households perceive their financial situation, can be more instrumental than objective material conditions in affecting overall financial wellbeing, health, educational attainment, productivity and decision-making (Layard and De Neve, 2023). Moreover, improvements in households' objective material conditions might not translate to improvements in subjective perceptions, revealing important information about adaptation mechanisms or spillovers to non-recipients (e.g. Seghieri, Desantis and Tanturri, 2006 in Europe; Jenkins, Sacker and Taylor, 2011 in Britain; Attah et al., 2016 in Ghana, Zimbabwe and Lesotho). In this paper, I analyse and further the understanding of MISs while giving subjective financial wellbeing the central place it merits as a key goal for researchers, policymakers and society as a whole.

The paper examines the case of Spain, a country that introduced a new MIS in 2020 (the *Ingreso Minimo Vital* or IMV). The IMV is an anti-poverty household-level measure available to Spanish residents above 23 years old with low income and wealth. It is a policy of great significance being the first non-contributory and non-categorical social benefit available at national level in the history of a country with persistently high levels of poverty and large regional differences. Up until the IMV introduction, the minimum income system in Spain was made up of 19 different regional MISs, which international institutions and academics assessed as having limited poverty alleviation capacity because of inadequate generosity, restrictive

eligibility criteria and low take-up (Arriba and Moreno, 2005; AIReF, 2019; European Commission, 2019; and Ayala *et al.*, 2021). The goal of the IMV is to homogenise this minimum income system, allowing beneficiaries to receive both the existing regional MISs and the common national-level IMV, which has more generous amounts and coverage than most regional schemes (see Figure 1 in Section 2.2). The policy aims to reach around 850,000 households in which 2.3 million individuals live.

The paper uses Eurostat survey data aggregated at the national level for the 2010-2022 period in a Synthetic Control Method (SCM) analysis as proposed by Abadie and Gardeazabal (2003) and Abadie, Diamond and Hainmueller (2010) as well as a Ridge Augmented SCM (RASCM) developed by Ben-Michael, Feller and Rothstein (2021), which is an extended version of the SCM that corrects for bias present in the SCM. The SCM has been coined as *'the most important innovation in the policy evaluation literature in the last fifteen years* ' by Athey and Imbens (2017: 9). It has the potential to present causal effects of policy interventions by effectively addressing challenges related to self-selection and to the need for behavioural assumptions. It achieves this by constructing a counterfactual made of a combination of control units.

The results show that, while the policy had no statistically significant effect on households' material conditions (i.e. the poverty rate, the poverty gap and mean income) for its first year and a half of existence, it did considerably improve how households perceive the evolution of their finances after two years and a half. The policy increased the balance between those saying their financial situation improved and those saying it deteriorated by a magnitude of between 10.1 and 14.6 points. Spain's new MIS acted as a lifeline in times of economic uncertainty during the Covid-19 and cost-of-living crises. The paper discusses several mechanisms explaining this differentiated impact of the policy, such as the lagged rollout of the IMV, the effectiveness of the adjustments made to the policy from 2022 as well as anticipation, placebo and positive spillover effects of the benefit. The findings stress the importance for practitioners to consider subjective measures when assessing MISs.

The paper makes two key contributions to the literature. First, this paper is the first to use innovative supervised machine learning in the form of a synthetic control method to evaluate the causal effect of an income support measure, thus addressing methodological issues of the existing literature. It is also the first to find a causal effect of the new Spanish IMV. Results could not only be useful to the Spanish administration but to countries with similar socio-

economic conditions (e.g. South and Eastern Europe and Latin America) (Castles *et al.*, 2010) and, more broadly, to any policy context sharing the specific IMV design elements. Second, this paper goes beyond the narrow emphasis on income measures found in the existing literature and examines the impact of a MIS on households' subjective financial wellbeing, which is key to comprehending the full impact of MISs like adaptation and spillover effects.

The remainder of the paper is structured as follows. Section 2 provides background information on the effects that minimum income schemes have on households' financial wellbeing. Section 3 outlines the methodology. Section 4 details the data. Section 5 presents the main results and robustness checks. Section 6 discusses the implications of the analyses and Section 7 concludes.

# 2 BACKGROUND

# 2.1 HOW CAN MINIMUM INCOME SCHEMES IMPROVE FINANCIAL WELLBEING?

Policymakers and researchers have traditionally focused on objective material conditions, looking at how household income changes with minimum income support. Income is indeed central to shaping financial wellbeing, as those who report higher levels of household income also report higher levels of overall financial wellbeing, even after accounting for differences in financial capability, personality characteristics and other influencers (e.g. Porter and Garman, 1990 in the US; Joo and Grable, 2004 in the US; Muir *et al.*, 2017 in Australia; West, Cull and Johnson, 2021 in Australia and Iramani and Lutfi, 2021 in Indonesia). Since MISs are targeted to households on the lower end of the income distribution, the financial status of households is typically assessed by comparing their income with the national average or median income, labelling households as 'poor' if their income is below a certain proportion of the median (Iramani and Lutfi, 2021).

Whether MISs improve households' income depends on various factors. In theory, giving cash to households should support them financially, improving their income and reducing poverty at the national level. However, the extent of this support depends on (i) the generosity of benefit amounts, which should lift recipients out of poverty; (ii) the capacity of the eligibility criteria to cover all those in need and (iii) the level of take-up by entitled individuals (Figari, Matsaganis and Sutherland, 2013 in the EU and Almeida, Poli and Hernández, 2022 in the EU).

Moreover, MISs can fail to improve households' real material conditions because of adverse labour supply effects. The classic economic model foresees that giving households cash based on their low means could incentivise the unemployed to remain unemployed, the working poor to stay in low-intensity and low-paid jobs as well as other workers (initially above the minimum income threshold) to reduce their labour supply to qualify for the benefit (e.g. Murray, 1984; Portney and Mead, 1990; Moffitt, 2016). This traps households in poverty rather than lift them out of precarious situations.

Several studies analyse how MISs affect households' objective financial wellbeing. Studies point to a reduction in poverty from MISs (e.g. Rodrigues, 2004; Canova, Piccoli and Spadaro, 2015; Frazer and Marlier, 2016; Notten and Guio, 2016 in Germany, Greece, Poland, United Kingdom; Gallo, 2021). Although these studies find effects that are larger on reducing more severe levels of poverty, such as poverty measured as the proportion of people who have a net income below 40% of the national median household income (e.g. Gorjón and Villar, 2019 in the Basque Country; Gallo, 2021 in Italy; Almeida, Poli and Hernández, 2022 in the EU) as well as on reducing the poverty gap, i.e. the average distance of those defined as poor to the poverty line (e.g. Behrendt, 2000 in Germany, Sweden and the UK; Brunori, Chiuri and Peragine, 2010 in Southern Italy; and Frazer and Marlier, 2016 in the EU).

However, the above-mentioned studies use ex-ante simulation models, which, as noted by Sutherland (2017), have strong assumptions about behavioural responses of beneficiaries, namely high levels of take-up and a preference for leisure over labour. These assumptions are contested empirically. Non-take-up of minimum income benefits by entitled individuals is a widespread issue in the European Union (EU) with a recent comparative study establishing that in Germany, Belgium, Finland and the Netherlands between 30% and 50% of the eligible population does not access the benefits (European Commission, 2022b). Moreover, individuals often prefer working to being jobless as they derive non-monetary gains from employment (e.g. social interactions, self-realisation or a sense of citizenship) (Lister, 2004) and as prolonged spells of unemployment are associated with mental and physical health problems (Jefferis *et al.*, 2011).

Moreover, these studies do not look at the effect on mean household income, which provides important complementary information to poverty measures. The latter are subject to changes in median income that can influence the interpretation of financial wellbeing. During an economic downturn like the Covid-19 pandemic, it might be that the median income falls so that the poverty threshold becomes lower. As a result, the number of people falling below this lower threshold is smaller and thus the poverty rate might be reduced or stay put even though the financial wellbeing of households has worsened due to the crisis. The mean disposable income can account for these changes in households' objective financial wellbeing. Hence, it is key to study the Spanish MIS in an ex-post analysis using a causal inference method and looking at different objective financial wellbeing indicators, namely the poverty rate, the poverty gap and mean income.

So far, I have discussed the centrality of household income to financial wellbeing and how a MIS could affect income. However, objective material conditions do not tell the whole story about households' financial wellbeing. Households' perceptions about their income matter too. Perceived changes in financial circumstances can be a stronger predictor of financial wellbeing than actual income changes as found by Brown, Taylor and Wheatley Price (2005) in Britain and by Winter *et al.* (1999) in Poland. It is thus important to also understand how MISs affect subjective financial wellbeing.

In theory, actual income changes through a MIS should be reflected in corresponding changes in households' perceived financial situation. Income support should improve subjective financial wellbeing via four main mechanisms: by allowing households to (i) improve their basic standards of living and expand choices in terms of consumption (including healthcare); (ii) improve their sense of control over finances as well as security and flexibility; and (iii) acquire goods, services and participate in activities that increase status within society (Lundberg *et al.*, 2010; Milligan and Stabile, 2011; Frijters and Krekel, 2021; and Simpson *et al.*, 2021).

However, it might also be that households' perceived change in their finances differs from their real situation, as their experiences are influenced by a range of factors (Dolan, Peasgood and White, 2008). First, improvements in objective living conditions can lead to short-lived and/or small improvements in subjective wellbeing because households' conception of the minimum satisfactory level of income increases over time depending on their new income or on that of others around them. To explain why countries increase their national income without corresponding improvements in happiness, Easterlin (1974) posited that people care about their income position relative to that of others.

Second, MISs can fail to improve subjective financial wellbeing despite material gains due to the stigma and shame attached to being poor and claiming social assistance. Since social assistance entitlements do not depend on contributions made in the past, many in society consider such benefits unfair and regard recipients as non-deserving (Moffitt, 1983; Currie, 2004). In this sense, society's attitudes towards social assistance in general and MISs in particular also matter for how income support affects subjective financial wellbeing.

Third, it might also be that the more stringent the conditionality measures attached to the receipt of social assistance (i.e. stricter job search requirements and harsher sanctions if conditions are breached), the lower the subjective financial wellbeing derived from such benefit (Haushofer and Shapiro, 2016 in Kenya; Lundberg, 2016 in the European Union; Davis, 2019 in the US; Wickham *et al.*, 2020 in the UK; Simpson *et al.*, 2021 in a review of studies in high-income countries). This is because greater conditionality reduces opportunities for recipients to pursue their own idea of a satisfying life and increases stress (Thornton and Iacoella, 2022).

Empirically, the literature on how MISs affect subjective financial wellbeing is very limited with only one Australian study looking at this explicit relation (Muir *et al.*, 2017) and finding that income support improves the perception of finances. However, while the literature has focused on how MISs affect the subjective wellbeing of recipients, the impact of MISs on households' perceived financial situation can go beyond direct beneficiaries in two main ways. First, following the idea that households' perception of their finances is affected by their relative rather than absolute income, the receipt of income support by some could degrade the relative income position of non-recipients causing jealousy and leading non-recipients to believe their financial situation has worsened through a mechanism of negative spillovers (Kassenböhmer and Haisken-DeNew, 2009 in Germany; and Kuhn *et al.*, 2011 in the Netherlands). It is worth noting that among small tight communities, the oppositive might take place: the higher income of some leads to more subjective wellbeing among neighbours because of empathy (e.g. Kingdon and Knight, 2007 in South Africa and Atsebi and Ferrer-i-Carbonell, 2022 in Tanzania).

Second, the mere announcement or introduction of an income support policy can lead to signalling effects and impact households' financial perceptions even before they have received the policy. There is a considerable body of literature noting how households regard policy announcements as a signal from the government and adjust their behaviours, expectations and confidence in economic conditions accordingly (OECD, 2010 in the UK; D'Acunto, Hoang

and Weber, 2019 in Germany; Lewis, Makridis and Mertens, 2019 in the US; Goldfayn-Frank, Kocharkov, and Weber, 2020 in Germany; and Melosi, Morita and Zanetti, 2022 in Japan). Hence, it is also key to study whether improvements in income through the Spanish MIS are reflected in perceived financial improvements, looking at this effect across the whole population.

## 2.2 HOW COULD THE SPANISH MIS IMPROVE FINANCIAL WELLBEING?

The theory points to different contradicting factors explaining how MISs affect households' actual income. Given these competing theories, I cannot predict the *overall* effect of the Spanish IMV on objective financial wellbeing indicators like the poverty rate, poverty gap and mean income, hence the need for an empirical analysis. On the one hand, the IMV has more generous amounts than most regional MISs (see Figure 1) and is expected to cover 247,000 more households than the regional MIS if fully implemented<sup>1</sup> (AIReF, 2022). As a result, the government expects to reduce the number of people living with less than 20% of the national median by 1.6 million (AIReF, 2019). Similarly, a microsimulation study estimated that the IMV would reduce the proportion of people living with less than 25% of the national median by 41% and that the corresponding poverty gap would be reduced by almost three quarters (a reduction from 4.6% to only 2.7% of the population) (Badenes Plá and Gambau-Suelves (2020). Thus, I could expect the IMV to reduce the poverty rate and gap and increase the average income in the country.

<sup>&</sup>lt;sup>1</sup> This figure excludes the Basque Country and Navarre. The figure also does not account for the increased demand for income support during the Covid-19 pandemic context.



Figure 1 – Comparison of Benefit Amounts Between Regional MIS and the IMV in 2020

Source: Own construction from Ministry of Social Rights (2022) Notes: Regional amounts are before the IMV introduction

Yet, on the other hand, the IMV amounts perform poorly when it comes to lifting people out of poverty measured at the 40% threshold, which is the indicator typically used to measure extreme poverty in high-income countries. Figure 2 shows the percentage of the national median household income that different types of households would reach if they benefitted from the IMV in 2020. Even if all types of households are brought closer to this threshold, only single-parent households manage to reach the 40% poverty threshold. Thus, even if the poverty gap is reduced and mean household income increased, it is less certain that the IMV will lower the poverty rate at the 40% threshold.



*Figure 2 – Percentage of Median Household Income Guaranteed by IMV Amounts by Household Type in 2020* 

Source: AIReF (2022)

In addition, the low take-up of the IMV cast even more doubt on the policy's overall effect on the national poverty rate, poverty gap and mean income. As of December 2020, i.e. half a year after its introduction, the IMV had only reached 0.46 million individuals or 20% of its initial objective (La Moncloa, 2021; Ministerio de Inclusion, Seguridad Social y Migraciones, 2022a). As of December 2021, the IMV had reached 0.82 million individuals (36% of its objective) and as of December 2022, it had reached 1.54 million (67% of its objective).

Moreover, the policy might generate employment disincentives failing to improve households' financial situation. Before 2022, the benefit withdrawal rate was relatively large and recipients only had to register as job seekers with the regional public employment services within 6 months *after* their application had been approved. This obligation was suspended altogether in April 2022 as part of the government's response to the cost-of-living crisis. Moreover, the policy was introduced during the Covid-19 crisis when access to the labour market was limited. Thus, an empirical analysis of the Spanish IMV is needed to establish its overall effect on objective financial wellbeing.

Similarly, I cannot predict from the competing theories on how MISs affect households' subjective financial wellbeing whether income improvements through the Spanish IMV will be reflected in perceived improvements in households' financial situation across the whole

population. Hence the need for an empirical analysis. On the one hand, it might be that the IMV leads to perceived financial improvements since it represents a stable source of income that allows beneficiaries to plan expenses, bringing stability and reducing stress (EAPN, 2021). Moreover, conditionality measures and stigma might not reduce subjective financial wellbeing in the Spanish case. As mentioned above, the IMV did not include stringent employment conditionality measures and Spain is the third EU country with the highest approval of cash benefits for socially excluded people with 73% of Eurobarometer respondents agreeing that the government should spend more on this area, compared to an average of 59% in the EU-27 (European Commission, 2022a). The IMV in particular received broad social support, with 83.4% of Spanish people supporting the policy in May 2020 (CIS, 2020).

On the other hand, it might be that the IMV leads to no improvements in how households perceive the evolution of their finances. Recipient households might rapidly adjust their perceptions to their new financial situation and Spain is not a particularly small tight community, meaning negative spillovers and envy among non-recipients might apply. It could also be that in the Covid-19 crisis context, the government's hasty introduction of the IMV within less than two months after being announced signalled to Spanish households that the situation is worse than they initially thought, thus making households feel more pessimistic about the evolution of their finances.

Thus, I conduct an empirical analysis to establish the overall effect of the IMV on households' objective and subjective financial wellbeing. Before turning to the analysis, I detail the identification strategy and data.

# **3** IDENTIFICATION STRATEGY

For this study, I use the Synthetic Control Method (SCM), which is a data-driven approach that allows for the construction of a counterfactual for the treated unit in the absence of the intervention through a weighted combination of control units. The SCM is a supervised machine learning tool, whose algorithm is fed data, trained to find patterns and ultimately, is set to predict observations based on the found patterns. This method was first introduced by Abadie and Gardeazabal (2003) in a study of the economic impact of terrorism in the Spanish Basque Country region and was further developed by Abadie, Diamond and Hainmueller (2010) in an analysis of the effect of a tobacco control programme in California.

The SCM is very fitting to estimate the effect of the IMV given that the method aims at estimating the impacts of interventions implemented at the aggregate level affecting a small number of large units. One notable advantage of the SCM, particularly when compared to a Difference-in-Differences (DiD) design, is its ability to circumvent the assumption that, in the absence of the intervention, the outcomes of interest in the treated unit would have followed the same trajectory as a single control country. Given the unique circumstances surrounding the Covid-19 pandemic and the cost-of-living crises, it is unlikely that countries have experienced similar trends in financial wellbeing since the implementation of the IMV in June 2020. The SCM allows for differences in the evolution of the outcomes of interests across countries since the control does not depend on a single country. Moreover, using a combination of control units has the potential to reproduce the characteristics of a treated unit better than a single comparison unit.

With the SCM, I create a 'Synthetic Spain without the IMV' made of different control countries selected for their similarity with Spain in terms of variables that predict objective and subjective wellbeing measures. I am then able to compare what would have happened to financial wellbeing in Spain in the absence of the IMV with the real changes in financial wellbeing in Spain with the IMV and attribute the effect to the policy intervention. The identifying assumption underlying the SCM is that the financial wellbeing of Spain without the IMV would have trended similarly to Synthetic Spain after June 2020. The only difference between real and Synthetic Spain must be the IMV.

Formally, in my SCM model, there is a sample of C + 1 countries so that c = 1, 2, ..., C + 1. The first country c = 1 is the treated unit (i.e. Spain) and c = 2, ..., C + 1 is a collection of untreated units not affected by the intervention (i.e. the donor pool of countries). In my analysis, I consider 11 EU countries as my donor pool. I choose EU countries since they have similar socio-economic and cultural characteristics to Spain and since Eurostat provides standardised and comparable data for these nations. In Section 3.2 below, I explain why I focus on 11 specific EU countries.

There is also a number T of time periods where I observe these units. The IMV introduction occurs at time  $T_0$  (i.e. June 2020) and is in place for the full post-treatment period starting with  $T_0 + 1$  where  $1 < T_0 + 1 < T$ . I perform a yearly analysis, where I study the period 2010-2021. I start the pre-intervention period in 2010 as this is the earliest date for which suitable data are available for all control countries and I end the study in 2021 since this is the latest

date for which income data is available for all countries. I also conduct a more fine-grained monthly analysis, which allows me to use more pre- and post-intervention data points, where I study the period December 2019 – December 2022 (see Section 4 for more data information and Section 5.1.4.2 for the rationale behind the monthly analysis).

There is no consensus in the literature as to a suitable number of pre-intervention periods, the only rule being that the longer the pre-intervention period, the more credible the model. While authors like Peri and Yasenov (2015), Adhikari and Alm (2016) and Tomasi (2022) use a relatively low number of pre-intervention periods (i.e. 4, 6 and 3 years, respectively), others such as Abadie, Diamond and Hainmueller (2010), Pang, Liu and Xu (2021) and Gilchrist *et al.* (2022) use larger periods (i.e. 19, 29 and 50 years, respectively). In my analyses, I have an 11-year and a 6-months pre-intervention period, which falls in line with the literature.

There are also several outcomes of interest  $Y_{ct}$  for each country, c and time, t. In my analyses, the major variables of interest are objective financial wellbeing measures (i.e. the poverty rate, the poverty gap and mean income) and a subjective wellbeing measure (i.e. households' perceived change in their financial situation) (see Section 4 for outcome definitions). Let the outcome of interest for country c at time t in the absence of the IMV be  $Y_{ct}^0$  and let  $Y_{1t}^1$  represent the outcome if the c-th country is exposed to the IMV so that Spain is represented as c = 1. I assume the intervention has no impact on the outcome before the IMV introduction such that  $Y_{1t}^1 = Y_{ct}^0$  for  $t < T_0 + 1$ .

The IMV's effect is described by  $\delta_{ct}$  and captures the difference between the observed and counterfactual financial wellbeing outcomes, which implies that  $\delta_{ct} = Y_{ct}^1 - Y_{ct}^0 = Y_{1t}^{Observed} - Y_{1t}^{Counterfactual}$ . Of these 2 outcomes, only one is observed. With the synthetic control estimator, I can model the other counterfactual outcome of Spain without the IMV. The synthetic control method uses an optimally chosen linear combination of the control countries that did not implement the IMV as a synthetic control unit.

To construct this control unit, the SCM sets weights for the countries in the donor pool and predictor variables. Let  $C \times 1$  be a vector of weights  $W = (w_2, ..., w_{C+1})'$  for each of the control countries.  $w_2, ..., w_{C+1}$  are non-negative  $w_c \ge 0$  and sum to one  $\sum_{c=2}^{C+1} w_c = 1$ . Thus, there is no extrapolation, meaning no country is given negative weights. Let  $K \times 1$  be a vector of weights  $V = (v_1, ..., v_k)'$  for each of the k predictors  $X_{1c}, ..., X_{kc}$  of financial wellbeing (see Section 4 for more information on predictors). Country weights W are chosen so that the resulting Synthetic Spain best resembles real Spain's pre-intervention values of financial wellbeing predictors. Optimal country weights  $W^* = (w_2^*, ..., w_{C+1}^*)'$  minimise:

$$\|X_1 - X_0 W\| = \sum_{k=1}^k v_k \left( X_{k1} - \sum_{c=2}^{C+1} w_c X_{kc} \right)^2$$
(1)

where  $v_k$  is a weight reflecting the importance given to the  $k_{th}$  predictor when measuring the discrepancy between predictors for real and Synthetic Spain and where  $X_{kc}$  is the value of the  $k_{th}$  predictor for country c.

As seen in Equation 1, the choice of country weights W depends on the choice of predictor weights V. Weights V are chosen to minimise the mean squared prediction error with respect to  $Y_{1t}$ , i.e. weights V are chosen so that the resulting Synthetic Spain best resembles Spain's pre-intervention financial wellbeing trends, which in turn informs the choice of country weights W:

$$\sum_{t=1}^{T_0} \left( Y_{1t} - \sum_{c=2}^{C+1} w_c^* (V) Y_{ct} \right)^2$$
(2)

To assess whether the estimated synthetic control accurately fits the path of the actual outcome for the treated unit in the pre-treatment period, I look at the mean squared prediction error (MSPE) between actual and Synthetic Spain's financial wellbeing outcomes during the pretreatment period. The MSPE should be small, especially when compared to the dispersion of the outcome variables, which I measure with the range. I also perform a visual inspection of the fit.

Subsequently, the difference between the real and synthetic unit, i.e. the average treatment effect on the treated unit (ATT) c = 1 (i.e. Spain) in period  $t = T_0 + 1, ..., T$  can then be estimated as:

$$\delta_{ct} = Y_{1t}^1 - \sum_{c=2}^{C+1} w_c^* Y_{ct}^0 \tag{3}$$

where  $Y_{1t}^1$  is financial wellbeing in Spain,  $Y_{ct}^0$  is financial wellbeing in the countries in the control group, and  $w_c^*$  is the optimally chosen weights for every country in the control group.

## 3.1 THE RIDGE AUGMENTED SYNTHETIC CONTROL METHOD

To test the robustness of the SCM results, I use a Ridge Augmented Synthetic Control Method (RASCM), which is an extension of SCM recently developed by Ben-Michael, Feller and Rothstein (2021). Abadie, Diamond and Hainmueller (2010) show that the SCM estimator is unbiased under the assumption that an exact balance, i.e. an excellent fit on pre-treatment outcomes, can be achieved. Exact balancing can only happen if the treated unit is in the convex hull of the control units, i.e. if the treated unit's pre-treatment values of outcome and predictor variables are within the set of control units' values. However, this probability decreases as the number of pre-intervention periods grows (Ferman and Pinto, 2021). For exact balancing weights to exist, the number of control units N must therefore be exponentially larger than the number of pre-intervention periods  $T_0$ . In most SCM settings this is not the case as usually N  $\sim T_0$ . My setting is no exception with N = 11 and  $T_0 = 11$  & 6, thus leading to potential bias and the need to corroborate results with a RASCM.

As a solution, Ben-Michael, Feller and Rothstein (2021) developed the Ridge Augmented Synthetic Control Method, which combines synthetic control weights with a regression adjustment for improved accuracy. The RASCM augments the SCM by using a different matching technique: it uses an outcome model to estimate bias in the SCM estimate when the pre-treatment match is not excellent and then uses this to de-bias the estimate. Ben-Michael, Feller and Rothstein (2021) propose the use of a ridge-regularised linear regression model as the outcome model. This approach can improve the pre-treatment fit by allowing for negative weights on some control units. Negative weights extrapolate outside of the convex hull, ensuring a much closer balance but resting more heavily on the assumption that the expected value of the outcome is approximately linear in the control outcomes. To minimise the extrapolation from the convex hull, the RASCM directly penalises the distance from nonnegative SCM weights. Formally, the RASCM estimator is:  $\hat{Y}_{1T}^{aug}(0) = \sum_{i=2}^{N} \hat{\gamma}_{i}^{aug} Y_{iT}$ , where the weights  $\hat{\gamma}^{aug}$  are a solution to:

$$\min_{\gamma} \frac{1}{2\lambda^{ridge}} \| \left( Y_{1T_0} - Y'_{iT_0} \gamma \right) \|_2^2 + \frac{1}{2} \| \left( \gamma - \hat{\gamma}^{scm} \right) \|_2^2 \tag{4}$$
Subject to  $\sum_{i=2}^N \gamma_i = 1$ 

Where  $\hat{\gamma}^{scm}$  are the SCM weights and  $\lambda^{ridge}$  is the hyperparameter which determines the amount of extrapolation (with the level of imbalance). Also,  $\|(Y_{1T_0} - Y'_{iT_0}\gamma)\|_2^2 \equiv (Y_{1T_0} - Y'_{iT_0}\gamma)'(Y_{1T_0} - Y'_{iT_0}\gamma)$  and  $\|(\gamma - \hat{\gamma}^{scm})\|_2^2 \equiv (\gamma - \hat{\gamma}^{scm})'(\gamma - \hat{\gamma}^{scm})$  are the 2-norm on  $\mathbb{R}^{T_0}$  and  $\mathbb{R}^{N-1}$ , respectively.

Following other studies using RASCM (e.g. Bouvet, Bower and Jones, 2022; Charotti, Palma and Santos, 2022; Esaka and Fujii, 2022; McGinty *et al.*, 2022; and Thom, 2022), I do not include covariates other than the pre-treatment outcome variable.

#### 3.2 CONSTRUCTION OF SYNTHETIC SPAIN

To respect the identifying assumption underlying the SCM, namely that the financial wellbeing of Spain without the IMV would have trended similarly to that of Synthetic Spain after June 2020, I need to carefully select countries in the donor pool.

The first threat to this identifying assumption comes from control units adopting similar interventions around the same time as the IMV introduction. Including any country in the donor pool that was treated in the period under investigation implies that the synthetic unit is not reproducing the potential outcome in the absence of treatment. If the synthetic control was made of these control units, the difference in post-treatment financial wellbeing between real and Synthetic Spain would be biased towards zero. Thus, my results would give a lower bound on the magnitude of the IMV effect.

Belgium, Finland, Italy, Latvia, Luxembourg and Portugal implemented policies comparable to the IMV, so I exclude these countries from the donor pool. I regard a measure implemented in another EU country as comparable to the IMV introduction if it either constitutes (i) an increase in the MIS amount or (ii) an increase in coverage by making eligibility rules less restrictive and if (iii) the changes are not a one-off measure and (iv) they happened around the same time as the IMV (March 2020-March 2021) (See Appendix 1 for a table summarising the policy changes implemented by the six excluded countries).

Since I study the Covid-19 crisis period, I also need to account for the fact that certain countries implemented considerably less (or more) income support measures than Spain at this time. Hale *et al.* (2021) developed an 'Economic Support Index', which measures countries' Covid-19-related income support and debt relief policies targeted at citizens<sup>2</sup>. I exclude countries that have an index consistently smaller (or larger) than Spain between March 2020 and March 2021 because were such countries to be part of Synthetic Spain, financial wellbeing would likely be smaller (larger) than it would have been in Spain in the absence of the IMV. This excludes Cyprus, Germany, Finland, Malta, Poland, Portugal, Slovenia and Sweden.

Another important threat to identification is the fact EU countries in the donor pool could have experienced the shock of Covid-19 very differently from Spain. If the donor countries that make up Synthetic Spain were less affected by Covid-19 and implemented less stringent lockdowns and regulations than in Spain, then post-intervention financial wellbeing would be greater than it would have been in Spain in the absence of the IMV, and vice-versa if countries were more affected than Spain. Hale *et al.* (2021) also developed a 'Stringency Index' measuring the strictness of lockdown policies by considering information on containment and closure policies<sup>3</sup> as well as public information campaigns. I exclude countries with a Stringency Index consistently below that for Spain between March 2020 and March 2021. This excludes Bulgaria, Denmark, Estonia, Finland, Latvia, Luxembourg and Sweden.

Finally, the presence of spillovers, i.e. the IMV affecting control countries, is another factor threatening the validity of the SCM assumption since it would mean Synthetic Spain trends differently than Spain without the IMV, leading to an under or overestimation of the policy effect. The IMV could affect other countries' financial wellbeing through changes in Spain's purchasing power and demand for foreign products. However, the indirect effect of the IMV

<sup>&</sup>lt;sup>2</sup> These policies include direct cash payments to people who lose their jobs or cannot work (only including payments to firms if explicitly linked to payroll/salaries), freezing of financial obligations for households (e.g. stopping loan repayments, preventing services like water stopping or banning evictions), economic stimulus spending and Covid-19 related aid spending to other countries.

<sup>&</sup>lt;sup>3</sup> These policies include closings of schools and universities, closings of workplaces, cancelling of public events, setting limits on gatherings, closing of public transport, orders to confine in the house, restrictions on internal movement between cities/regions and restrictions on international travel.

on other countries' financial wellbeing through imports/exports is likely to be very small. The IMV could also impact other EU countries' financial wellbeing by attracting EU residents in search of social benefits. Yet, there is limited evidence on the existence of 'welfare tourism' in the EU (Dustmann, Frattini and Halls, 2010; European Commission, ICF GHK and Milieu Ltd, 2013).

I end up with a donor pool of 11 EU countries to construct Synthetic Spain made of Austria, Croatia, Czech Republic, France, Greece, Hungary, Ireland, Lithuania, the Netherlands, Romania and Slovakia. This is in line with Abadie, Diamond and Hainmueller's (2015) recommendation of limiting the donor pool to countries similar to the one affected by the intervention to avoid the risk of overfitting and thus artificially matching the characteristics of the treated unit by combining idiosyncratic variations in a large sample of unaffected units.

# 4 DATA

I use national-level panel data from Eurostat for the period 2010 - 2022. The panel data is unbalanced.

#### 4.1 OUTCOME VARIABLES

#### 4.1.1 Objective Financial Wellbeing Measures

I investigate the *poverty rate*, i.e. the proportion of people who have a net income below 40% of the national after-tax and transfers median household income. This is a yearly variable constructed from Eurostat's EU statistics on income and living conditions (EU-SILC) survey. I use a relative rather than absolute poverty measure this is the most common understanding of poverty in European policymaking and academia. I take the 40% rather than the 60% poverty threshold since the IMV amounts are closer to the 40% threshold (See Figure 2) and the literature has found bigger effects on this measure, as explained in Section 2. This is the lowest poverty threshold provided by Eurostat.

I also include the *poverty gap*, i.e. the difference between the median equivalised disposable income of people below the 40% poverty threshold and the poverty threshold, expressed as a percentage of the poverty threshold. This is a yearly variable constructed from Eurostat's EU-SILC survey. I focus on this measure since the literature has found bigger effects on this

measure and the IMV amounts are more likely to bring people closer to the 40% threshold rather than take them above that, as explained in Section 2.

I also look at *mean disposable household income*, which is the sum of all the disposable income of all households divided by the number of households in the country. Disposable household income includes all income from work (employee wages and self-employment earnings), private income from investment and property, transfers between households and all social transfers received in cash, including old-age pensions. I look at mean disposable income on top of poverty measures since the latter are subject to changes in median income that can influence the interpretation of financial wellbeing.

#### 4.1.2 Subjective Financial Wellbeing Measure

I examine the *perceived change in households' financial situation*. This is a monthly variable from Eurostat's Consumer Survey, which asks respondents how the financial situation of their household has changed over the past 12 months. Respondents can either answer (i) got a lot better; (ii) got a little better; (iii) stayed the same; (iv) got a little worse; (v) got a lot worse; or (vi) don't know. The variable is a balance, i.e. the difference between the percentage of respondents giving positive and negative answers. A negative balance means more people are saying their financial situation deteriorated than people saying it improved and vice-versa.

There are two potential caveats with this measure. First, while the question asks about the evolution of the financial situation of the whole household, the response is given by one household member alone. Yet, the opinion of one household member might differ from that of other members depending on (i) asymmetries in the allocation of resources within the household or in the information about household finances that different members possess; (ii) the personalities of different members, with women being on average more pessimists than men when it comes to their economic situation (Jacobsen, Lee and Marquering, 2008 in Europe and the USA); and (iii) different amounts of pressure households members feel to answer in a way that conforms with societal demands to succeed or that would please the interviewer (Bertrand and Mullainathan, 2001 and Bryman, 2016).

However, in my analysis, I assume that the answer given by the survey respondent reflects the perception of the entire household since the measure I study is not *cardinal*, asking respondents by how much the financial situation of the household has changed, but rather *ordinal*, merely asking whether their situation has changed. An ordinal measure leaves less room for

interpretations of the evolution of household finances and thus is less likely that significant differences will emerge among household members. Moreover, Eurostat's Consumer Survey targets respondents to achieve a representative sample of the population in terms of sex, age, education, income and occupation so that when the measure is aggregated across the whole national population, different approaches to the question across the survey respondents can be expected to cancel out.

Second, perception measures could be subject to the difficulty of inter-household comparisons of mental states (Bertrand and Mullainathan, 2001): for the same increase in household income, one household might perceive it as a big improvement, while another might consider its situation has only improved a little<sup>4</sup>. However, the specific measure I study limits this concern as it adds all positive responses and subtracts them from all negative ones. Moreover, numerous academics such as Winter *et al.* (1999), Di Tella and MacCulloch (2006) or Kahneman and Krueger (2006) have evidenced that self-reported subjective wellbeing is a stable concept that can be measured reliably across people.

#### 4.2 PREDICTOR VARIABLES

I choose fourteen predictors based on what the literature has found to explain financial wellbeing in a country. Predictors are:

- (i) *pre-intervention values of the poverty rate* (Kaul *et al.*, 2021);
- (ii) pre-intervention values of the poverty gap (ibid);
- (iii) pre-intervention values of the mean disposable income (ibid);
- (iv) pre-intervention values of the perceived change in households' financial situation (ibid);
- (v) *difficulty making ends meet*, i.e. percentage of households reporting having great difficulty making ends meet (Salignac *et al.*, 2020);
- (vi) Gross Domestic Product per capita, i.e. the ratio of the value of total final output of goods and services produced by an economy to the average population in a given year (Blank et al., 1993);
- (vii) *Gini coefficient* measuring the dispersion of income within a country and thus inequality (Hoynes, Page and Stevens, 2005);

<sup>&</sup>lt;sup>4</sup> A household could also consider a certain income increase an improvement while another household would regard the same increase as a worsening. However, such a situation seems highly unlikely.

- (viii) *low educational attainment*, i.e the population share with less than primary or just primary and low secondary education (ISCED 0-2) (Andriopoulou and Tsakloglou, 2011);
- (ix) *high educational attainment*, i.e. population share with tertiary education (ISCED 5-8) (*ibid*);
- (x) social protection spending, i.e. total government general expenditure on sickness and disability, old age, survivors, family and children, unemployment, housing, social exclusion and social protection related research and innovation, as a share of GDP (Danziger and Gottschalk, 1995);
- (xi) *health spending*, i.e. total government general expenditure on medical products, appliances and equipment, outpatient services, hospital services, public health services and health-related research and innovation, as a share of GDP (*ibid*);
- (xii) single-parent households, i.e. the share of households with one adult and dependent children (Hoynes, Page and Stevens, 2005);
- (xiii) childcare use, measured by the percentage of children aged between 3 years old and the minimum compulsory school age, who spend 30 hours or more per week in formal childcare (Grigoli, Koczan and Topalova, 2018); and
- (xiv) *unemployment rate*, i.e. the share of people aged 15 to 74 who are not employed, currently available for work and actively seeking work as a percentage of the labour force (*ibid*).

In the monthly analysis, I only include the *pre-intervention values of the perceived change in households' financial situation* and the *unemployment rate* as the only predictors because of the limited availability of monthly data for countries in the donor pool.

# 5 <u>RESULTS</u><sup>5</sup>

# 5.1 YEARLY ANALYSIS

## 5.1.1 Composition of Synthetic Spain

In the first step, the SCM algorithm assigns weights to each of the predictors that can range from 0 to 1 but must add up to 1, such that the variables that can most accurately predict the trend in financial wellbeing of Spain before the treatment receive the highest weights. Table 1

<sup>&</sup>lt;sup>5</sup> All analyses are performed using the 'synth' and 'augsynth' packages in R.

displays the weights assigned to the predictor variables used to construct Synthetic Spain for each of the models. Each of the four models corresponds to a different outcome variable.

Looking at the poverty rate, the most important predictors are the pre-treatment average of the poverty rate and of the poverty gap as well as the Gini coefficient, which together account for 71.6% of all weights. Looking at the poverty gap, the pre-treatment averages of the poverty gap and poverty rate as well as the population share reporting having great difficulty making ends meet are most important, accounting for 65% of weights. In terms of mean household income, weights are more spread across all predictors. In the perceived change in financial situation model, it is the pre-treatment average of the perceived change in households' financial situation and the population share with high educational attainment, which are the most important predictors, accounting for 73.9% of weights.

	Model 1 – Poverty rate	Model 2 – Poverty gap	Model 3 – Mean income	Model 4 – Perceived change in financial situation
Poverty rate	34.9	15	0.4	0.1
Poverty gap	20.9	18.8	1.3	3.8
Mean income	0	8	10.3	0
Perceived change in financial situation	7.4	14.6	0.1	30.1
Difficulty making ends meets	3.5	31.2	14.8	1.7
GDP per capita	0.3	6.9	0.7	0.2
Gini coefficient	15.8	1.6	6.6	0.5
Social protection spending	4.8	0	1.8	12.5
Health spending	0	0.8	9.5	5.2
Low education attainment ISCED 0-2	3.3	0.2	3.6	1.7
High education attainment ISCED 5-8	0	1.3	19	43.8
Single-parent household	6.7	1.5	2.3	0.2
Childcare use	2	0	16.3	0.1
Unemployment rate	0.2	0	13.3	0.1
MSPE	0.01	1.43	366	7.75
Range (max-min)	12.8	50.5	30,522	89.26

*Table 1 – Weights (in Percentages) Assigned to Predictor Variables Used to Construct Synthetic Spain* 

Note: Although some predictors are presented as having zero weight, the real value is not exactly zero but close to zero (e.g.  $10^{-5}$ ).

Next, the algorithm finds Synthetic Spain as a linear combination of countries in the donor pool, such that Synthetic Spain matches the values of the predictor variables with the highest predictive power for real Spain as closely as possible and the Mean Squared Prediction Error (MSPE) of the outcome variables before the intervention is minimised. Table 2 displays the

weights assigned to each control country for each model. Greece, the Netherlands, Lithuania and Romania, play an important role in the construction of Synthetic Spain.

	Model 1 – Poverty rate	Model 2 – Poverty gap	Model 3 – Mean income	Model 4 – Perceived change in financial situation
Austria	0	15.3	0	0
Croatia	0	0	0	0
Czech Republic	0	0	0	0
France	12.4	0	26.6	0
Greece	21.9	15.7	42.8	24.4
Hungary	0	0	0	0
Ireland	0	0	9.3	22.9
Lithuania	0.1	15.5	12.2	1
Netherlands	5.9	16.8	9.1	52.7
Romania	59.7	36.7	0	N/A
Slovakia	0	0	0	0

Table 2 - Weights (in Percentages) Assigned to Countries Used to Construct Synthetic Spain

Note: Although some predictors are presented as having zero weight, the real value is not exactly zero but close to zero (e.g.  $10^{-5}$ ). These countries receive such small weights because their similarity in predictors to Spain is smaller than for countries with larger weights.

Table 3 presents the pre-treatment sample means of predictors for Spain, Synthetic Spain and the countries in the control group. The differences in the averages between Spain and Synthetic Spain are mostly minor for all predictors and all models. Notable exceptions include the population share with low education attainment (in all models) and high attainment (Models 1 and 2), the unemployment rate (Models 1, 2 and 4), childcare use (Model 1, 2 and 4), the poverty rate (Model 3 and 4), mean income (Model 1 and 4), the population share having difficulties to make ends meet (Model 1 and 3), GDP per capita (Model 1 and 4), the perceived change in financial situation (Model 3), the poverty gap (Model 3) and the Gini coefficient (Model 4).

However, where these variables are different from the average value in Spain, the predictors have weights under 5% in their corresponding models, except for the population share having difficulties making ends meet in Model 3 with a weight under 15%. Thus, these different predictors do not considerably impact the construction of the synthetic control.

		Synthetic	Synthetic	Synthetic	Synthetic	Average of 11
	Spain	Spain	Spain	Spain	Spain	Control
		Model 1	Model 2	Model 3	Model 4 <sup>a</sup>	Countries
Poverty rate	9.88	9.84	8.32	6.17	4.68	5.71
Poverty gap	34.42	33.32	33.61	27.12	31.26	28.55
Mean income	16,502	8,727	12,295	16,702	21,885	14,046
Perceived change in financial situation	-20.90	-20.83	-17	-30	-21.69	-17.27
Difficulty making ends meets	13.20	19.72	15	19.93	13.35	13.05
GDP per capita	25,280	20,065	24,088	27,246	34,568	26,092
Gini coefficient	33.90	33.14	32.04	31.60	28.90	29.58
Social protection spending	17.72	15.76	15.80	19.40	16.60	16.28
Health spending	6.35	5.33	5.80	6.33	6.87	6.45
Low education attainment ISCED 0-2	43.46	28.73	23.71	25.48	28.18	22.17
High education attainment ISCED 5-8	32.13	19.81	23.91	30.50	31.30	25
Single-parent household	3.05	2.60	3.81	3.85	4.06	3.79
Childcare use	42.40	22.50	28.37	42.53	21.84	42
Unemployment rate	20.05	9.68	7.52	14.30	10.34	9.37

# Table 3 – Descriptive Statistics (Means) for Predictor Variables Before the Intervention

Notes: The table presents the predictor variable mean values for real Spain, the synthetic control unit and the average of 11 control countries between 2010 and 2020

I consider 2020 a pre-intervention period because this allows me to account for the shock caused by the Covid-19 crisis in the synthetic control construction

<sup>a</sup>: The model excludes Romania from the donor pool as data for the outcome variable is unavailable for the full post-intervention period

The similarity in most predictors' means before the intervention points to a good pre-treatment fit between Spain and Synthetic Spain across the different models. Moreover, as shown in Table 1, the MSPE of all models is low when compared to the ranges of the outcome variables. The good fit of financial wellbeing measures before the IMV was introduced is also corroborated by a visual inspection of Figure 3 below.

# 5.1.2 Main Results on the IMV Effect on Financial Wellbeing

*Figure 3 – Evolution of Financial Wellbeing Outcomes & Differences in Outcomes between Spain and Synthetic Spain* 





Figure 3 displays the trends in financial wellbeing outcomes for Spain and Synthetic Spain (left-hand panel) as well as the difference in these outcomes between Spain and Synthetic Spain (right-hand panel). All financial wellbeing measures in Spain follow approximately the same trend: a deterioration or stagnation during the Great Recession years, followed by a recovery until the Covid-19 pandemic hit. Financial wellbeing worsened in 2020 to then rebound in 2021. In this analysis, I am interested in investigating whether financial wellbeing in 2021 improved thanks to the IMV, i.e. whether financial wellbeing is higher in real Spain than in Synthetic Spain in the absence of the minimum income support.

Between 2010 and 2014, the poverty rate increased steadily, peaking at 11.2% of the population in 2014. Since then, the poverty rate declined until 2019, reaching 9.5%. In 2020, this rate increased again as could be expected in the context of the Covid-19 pandemic, when people either lost their jobs or saw their incomes decrease. However, by 2021, the poverty rate in Spain had decreased again and stood at 8.9%. The poverty rate of Synthetic Spain in the same year was 8.85%, indicating that the IMV could have increased the poverty rate by 0.05 percentage points. In section 5.1.3. below, I test whether the results are statistically significant.

The poverty gap follows a more erratic path, continuously increasing and decreasing between 2010 and 2020. Nonetheless, the overall trend shows an increase in the gap from 2010, peaking at 36.2% in 2014, followed by a decrease, reaching 31.1% in 2019. Similar to the poverty rate results, the poverty gap increased during the Covid-19 crisis and then decreased again, standing at 29.8% in 2021. The poverty gap in Synthetic Spain in 2021 was 35%. Thus, it would seem that the IMV reduced the poverty gap in Spain by 5.2 percentage points.

The mean household income in Spain decreased between 2010 and 2013, as can be expected during the Great Recession years. The Spanish mean income reached a minimum of 15,405 euros in 2013 to then recover and keep increasing until 2019. Mean income decreased slightly in 2020 to then pick up again at 19,160 euros in 2021. The mean household income in Synthetic Spain in 2021 was 19,024. It would seem that the IMV increased mean income in Spain by 136 euros.

Finally, the perceived change in the financial situation of households follows a similar pattern. The difference between those households saying their financial situation improved over the past 12 months and those saying it deteriorated, decreased during the crisis years to reach a minimum balance of -37.30 in 2013. The balance then picked up to reach -6.45 in 2019. Likely because of the Covid-19 pandemic, Spanish households became more pessimistic about the evolution of their finances with an increasing proportion of respondents saying their situation worsened compared to those saying it improved. By 2021, the financial situation change balance stood at -20.13. The balance for Synthetic Spain in the same year was -20.30, indicating that the IMV could have increased the balance by 0.17 points.

In sum, it appears that while the Covid-19 crisis in 2020 reduced financial wellbeing, both objective and subjective measures, by 2021, objective financial wellbeing indicators had recovered to pre-crisis levels. On the contrary, subjective financial wellbeing took longer to

cover and was still deteriorating in 2021. The more persistent deterioration in the perceived financial situation might be influenced by the uncertainty caused by the pandemic, which could have contributed to a sense of stress and lack of control over finances. The results seem to lend support for the literature positing that objective and subjective financial wellbeing measures move in different directions.

In terms of the effect of the IMV, while it seems that the policy slightly increased the poverty rate, it also seems to have had more virtuous effects on the other financial wellbeing indicators: the IMV could have reduced the poverty gap, increased the mean household income and made household feel less pessimistic about the evolution of their finances. I now turn to establish whether these results are statistically significant or have been achieved by chance.

#### 5.1.3 Statistical Significance of Main Results

In a SCM, commonly used large-sample inferential techniques cannot be used to assess the statistical significance of the results since the number of control groups is small. The standard way to assess the significance of effects in a SCM is by calculating pseudo *P*-values obtained through permutation-based placebo experiments. These placebo experiments iteratively estimate the placebo treatment effect for each unit in the donor pool by falsely assuming that these units introduced the IMV. This tests whether any post-intervention difference in financial wellbeing between the synthetic unit and actual Spain could be driven entirely by chance or by a failure to accurately reproduce the counterfactual development of financial wellbeing. If this difference is large for real Spain compared to the placebos, this could be suggestive of an effect rather than as a traditional null hypothesis-based inference, Firpo and Possebom (2018) found that this method performs well compared to other test-statistics in terms of size, power and robustness.

To obtain the pseudo-P-values, I repeat the analysis for the 11 countries that did not implement the IMV. I then calculate the post/pre mean squared prediction error (MSPE) ratio, that is the ratio of the root MSPE after and before 2020. The ratio gives the difference between the financial wellbeing of a unit and its synthetic control before and after treatment. A higher ratio means a small pre-treatment prediction error (a 'good' synthetic control) and a high posttreatment MSPE (a large difference between the unit and its synthetic control after the intervention). Figure 4 displays the ratios for each of the countries in the donor pool and Spain in each of the four models (left-hand panel). I exclude all placebo cases with a pre-period MSPE exceeding two times the treated unit's pre-period MSPE, i.e. all those countries with a 'bad' synthetic control. Figure 4 also shows a 'spaghetti plot' displaying the difference in financial wellbeing outcomes of each country in the donor pool and in Spain, which is represented by the bold line (right-hand panel).

*Figure 4 – Post/Pre-intervention Mean Squared Prediction Error for Spain and Control Countries & Spaghetti Plots* 



- Control units - ES





Model 4 - Perceived change in household financial situation



The pseudo *P*-values is given by the fraction of the estimated post/pre MSPE ratios that are as large as the one estimated for Spain. The ratios are 7/9, 3/10, 7/7 and 11/11 for Models 1, 2, 3 and 4, respectively. Thus, the probability of obtaining an estimated effect on financial wellbeing at least as great as Spain was 77.8%, 30%, 100% and 100% in Models 1, 2, 3 and 4, respectively if the intervention was reassigned at random to the other countries. Adhikari

(2022) notes that if the placebo experiments create placebo treatment effects of magnitude greater than the one estimated for the treated unit in more than 10% of the placebo experiments (i.e. if the corresponding pseudo-P-value is greater than 0.1), then one can conclude that there is no evidence of an effect of the policy in the treated unit. Thus, I cannot conclude any statistically significant effect of the IMV in 2021 on either objective or subjective financial wellbeing measures in Spain.

#### 5.1.4 Robustness Checks

I perform three different types of tests to check that the SCM analysis has been able to correctly reproduce the financial wellbeing trends in Spain in the absence of the IMV. I perform a Ridge Augmented Synthetic Control Analysis, 'leave-one-out' tests and in-time placebo tests.

# 5.1.4.1 Ridge Augmented Synthetic Control Analysis

As explained in Section 3.2, the validity of the SCM results depends on the assumption that an excellent pre-treatment fit can be achieved. Otherwise, results might be biased. However, a visual inspection of Figure 3 determines that there are some gaps in the pre-treatment fit of outcome variables between Spain and Synthetic Spain. Thus, I conduct the same analysis using the RASCM with no covariates. The method corroborates the insignificance of results found with the SCM.

Figure 5 shows how the pre-treatment fit has improved considerably in all models with the RASCM compared to the SCM. Differences in outcomes between Spain and Synthetic Spain are mostly similar to those obtained with the SCM, except that the magnitude is now greater. It appears that the IMV decreased the poverty rate and gap as well as increased mean income and improved how households perceive the evolution of their finances.

Table 4 shows the magnitude of the differences in financial wellbeing outcomes between Spain and Synthetic Spain after the IMV introduction, i.e. the estimated average treatment effects of the IMV. The Spanish poverty rate and poverty gap seem to be 4.7 and 7.25 percentage points lower, respectively, than they would have been without the IMV. The mean income and the perceived change in the financial situation are also 459.75 euros and 6.29 points higher than they would have been with no IMV.

However, Table 4 shows how none of these effects are statistically significant at conventional levels since the corresponding P-values of these estimates are all above 0.1. It can be concluded that, in 2021, the IMV had no statistically significant effect on either objective or subjective financial wellbeing. Table 5 presents the weights given to the donor countries used to construct Synthetic Spain with the RASCM.

*Figure 5 - Evolution of Differences in Financial Wellbeing Outcomes between Spain and Synthetic Spain* 

Model 1 – Poverty rate

Model 2 – Poverty gap



Model 3 – Mean income

Model 4 - Perceived change in financial situation



Estimate Model 1 – Poverty rate	P-Value Model 1 – Poverty rate	Estimate Model 2 – Poverty gap	P-Value Model 2 – Poverty gap	Estimate Model 3 – Mean income	P-Value Model 3 – Mean income	Estimate Model 4 – Perceived change in financial situation	P-Value Model 4 – Perceived change in financial situation
-4.70	0.69	-7.25	0.52	459.75	0.42	6.29	0.25

Table 4- Estimated ATT for Post-intervention Years & Corresponding P-values

*Table 5 - Weights Assigned to Countries Used to Construct Synthetic Spain Using the RASCM* 

	Model 1 – Poverty rate	Model 2 – Poverty gap	Model 3 – Mean income	Model 4 – Perceived change in financial situation
Austria	-3.76	0.02	-0.06	-0.57
Croatia	-0.86	0.28	0.21	0.63
Czech Republic	-0.33	-0.25	-0.1	0.94
France	0.20	0.47	0.13	0.15
Greece	1.78	0.10	0.44	-0.14
Hungary	-0.35	0.06	-0.05	-0.11
Ireland	-0.43	0.04	0.27	-0.44
Lithuania	-0.27	0.23	0.17	-0.21
Netherlands	5.53	0.02	0	0
Romania	-1.44	0.02	-0.65	0
Slovakia	0.92	0	0.64	0.76

## 5.1.4.2 Other robustness checks

The ability of Synthetic Spain to correctly reproduce financial wellbeing in real Spain without the IMV depends on the choice of units in the donor pool and the choice of predictors of the outcome variables. If removing a donor country or predictor leads to a good pre-treatment fit but to different results post-intervention, my results could be driven by a post-intervention shock to a donor country or a predictor variable so that, in the absence of the IMV, Spain would not have trended as Synthetic Spain. This is a nonnegligible possibility in my analysis since I study the Covid-19 crisis period.

To test for the assumption that results are not driven by the choice of control countries or predictors, I conduct 'leave-one-out' tests. The tests consist in taking out from the sample each of the control countries and predictors one at a time (Gilchrist *et al.*, 2022). If the exclusion of control units or predictors leads to estimates before and after the intervention that are close to those using all donor pool countries or all predictors, then results can be considered robust

(Abadie, 2021; McClelland and Mucciolo, 2022). The 'leave-one-out' test for predictors has no impact on the direction and (in)significance of the effects found. In the 'leave-one-out' test for countries, the exclusion of Romania in the poverty rate and gap analyses and of the Netherlands in the perceived financial situation, do affect the direction and size of the estimates (See Appendix 9.2 for graphs). This is expected as these are the countries that are given large weights in my analysis. However, the insignificance of the 'leave-one-out' estimates are the same as when all countries are used. Moreover, the pre-treatment fit when excluding these countries is worsened significantly implying the difference in estimates is not driven by a shock to the excluded countries, which would confound my results when including this country, but rather by the fact that Synthetic Spain cannot be accurately constructed without those countries.

The validity of my SCM results also rests on the assumption that the only treatment effect found happened after the IMV intervention. If an effect is found before the intervention, this would call the model's predictive power into question, implying my post-intervention results could be biased. I thus need to check that there are no unusually large and statistically significant treatment effects before 2021 when the treatment was not implemented relative to the effect found for 2021 (Pang, Liu and Xu, 2021). I perform a time placebo test by reassigning the treatment status to each other year in the pre-intervention period. I also test for the statistical significance of these results. No placebo intervention yields any effect, except for a few results having effects larger than those found for 2021 and having a relatively good pre-treatment fit, but which turned out to be insignificant (See Appendix 9.3 for graphs).

From the SCM and RASCM analyses, I conclude that the IMV had no statistically significant effect on households' real and perceived financial situation. However, it might be that the results are insignificant because I am not accounting for the full effect of the IMV. First, I use the whole of 2020 as a pre-intervention period. This allowed me to account for the Covid-19 effect in the creation of the synthetic control, which is important to accurately construct Synthetic Spain to mimic as closely as possible real Spain. However, it means I might be losing some of the shorter-term effects of the IMV between June and December 2020. To solve for this potential underestimation of the IMV effect, I would need monthly data that includes June-December 2020 in the post-intervention period.

Second, I only study one post-intervention period (i.e. 2021). This might be too little time to be able to see any effect of the IMV on national financial wellbeing outcomes, especially considering the lagged rollout of the policy. If, as of December 2021, only 0.82 million

individuals had benefited from the IMV out of the 2,3 million the government hoped to reach, by December 2022, the IMV had reached 1.54 million individuals (La Moncloa, 2021; Ministerio de Inclusion, Seguridad Social y Migraciones, 2022a). To analyse the full impact of the IMV, I would need data that goes beyond 2021.

In the next section, I perform the same analysis but using monthly data between December 2019 and December 2022, which allows me to have 31 post-intervention points as well as to analyse the effect of the IMV during its first few months of implementation whilst accounting for the Covid-19 shock. Since monthly income data from surveys is not available, I can only conduct this monthly analysis for the subjective financial wellbeing measure.

## 5.2 MONTHLY ANALYSIS

#### 5.2.1 Composition of Synthetic Spain

Table 6 presents the pre-intervention sample means of predictor variables for Spain, Synthetic Spain and the 10 countries in the control group. The similarity of both predictors means before the intervention shows a good pre-treatment fit between Spain and Synthetic Spain. Moreover, as can be seen in Table 7, the MSPE is low when compared to the range of the outcome variable showing that the fit of the perceived change in households' financial situation before June 2020 is good. This is also corroborated by a visual inspection of the pre-treatment fit in Figure 6 below.

*Table 6 – Descriptive Statistics (Means) for Predictor Variables Used to Construct Synthetic Spain* 

	Spain	Synthetic Spain – Perceived change in financial situation	Average of 10 Control Countries <sup>a</sup>
Perceived financial situation	-13.39	-13.36	-4
Unemployment rate	14.63	14.63	6.74

Notes: The table presents the predictor variable mean values for real Spain, the synthetic control unit and the average of 10 control countries between December 2019 and May 2020.

<sup>a</sup>: The model excludes Romania from the donor pool as data for the outcome variable is unavailable for the full post-intervention period.

	Perceived change in financial situation
Perceived financial situation	0.40
Unemployment rate	99.6
MSPE	8.90
Range (max-min)	70.6

*Table* 7 – Weights (in Percentages) Assigned to Predictor Variables Used to Construct Synthetic Spain

Surprisingly, Table 7 reveals that the unemployment rate is given almost all the weight in the construction of Synthetic Spain. However, the almost identical value of the pre-treatment perceived change in households' financial situation for Spain and Synthetic Spain shown in Table 6 implies that the constructed Synthetic Spain would be able to mimic well the perceived change in the financial situation of households after the IMV is introduced. In Section 5.2.2., I prove that excluding the unemployment predictor does not affect results.

The resulting Synthetic Spain is a weighted average of all countries with Greece and Hungary playing the greatest roles. Table 8 displays the weights assigned to each control country.

	Perceived change in financial situation
Austria	1.3
Croatia	1.5
Czech Republic	0.9
France	1.6
Greece	75.1
Hungary	14.9
Ireland	1.2
Lithuania	1.6
Netherlands	1.1
Slovakia	1

Table 8 - Weights (in Percentages) Assigned to Countries Used to Construct Synthetic Spain

#### 5.2.2 Main Results on the IMV Effect on Financial Wellbeing



*Figure 6 – Evolution of Subjective Financial Wellbeing & Differences in Outcomes between Spain and Synthetic Spain* 

Figure 6 displays the trend in the perceived change in financial situation for Spain and Synthetic Spain. From March 2020, as the Covid-19 crisis triggered strict lockdowns, an increasing number of households became more pessimistic about the evolution of their financial situation. The perceived change in financial situation balance reached -24.6 in June 2020. From then onwards, the situation seemed to stabilise and even improved, peaking at a balance of -14 in October 2021. This improvement in the subjective financial situation change of households is likely because of the governments and the EU's support packages to households, workers and businesses. Since then, the subjective evolution of financial wellbeing of households started to deteriorate again and the balance reached a new low of -35.7 in July 2022. This corresponds to the start of the cost-of-living crisis as supply disruptions given ongoing Covid restrictions in large exporting countries, increased global demand after the lifting of restrictions in most economies and, later from February 2022, the disruption to the supply of gas and animal feed following Russian's invasion of Ukraine, meant prices rose for households, decreasing their purchasing power and sense of control over finances.

In terms of how Spain with the IMV fared compared to Synthetic Spain without the IMV, the IMV seems to have softened the blow to households' perceived change in their financial situation during the Covid-19 and cost-of-living crises. The evolution of the financial situation of households in Spain with the IMV outperforms that of Synthetic Spain, especially from July 2021. This can be seen from the departing curves in the left-hand panel of Figure 6 and from

the increasingly positive differences in the perceived change in finances between Spain and Synthetic Spain shown in Table 9 below. The subjective financial wellbeing measure presented here gathers information on how households perceive that their finances have changed *over the past 12 months*. As such, the large IMV effect observed from July 2021 reflects any perceived changes to household finances having occurred between July 2020 and July 2021, which coincides with the introduction of the IMV in June 2020.

The average total IMV effect between the post-intervention period of June 2020 and December 2022 is an increase in the difference between respondents perceiving an improvement in their finances and those perceiving a deterioration of 10.1 points. The size of this policy effect over two and a half years is non-negligible since it corresponds to over a fifth of the improvement in the perceived change in financial situation balance that took place during the economic recovery of the financial and debt crises over a six-and-a-half-year period between November 2012 and June 2019<sup>6</sup>. In Section 5.2.1. below, I investigate if this result is statistically significant.

<sup>&</sup>lt;sup>6</sup> In November 2012, the perceived change in financial situation balance reached a record low of -45.7. It took over six and a half years for household financial confidence to reach pre-crisis levels, when the balance peaked at -0.7 in June 2029. This corresponds to an improvement in the balance between those saying their financial situation improved and those saying it deteriorated of 45 points.

	Time period	Spain	Synthetic Spain	Difference
	2019-12	-8.4	-9.4	1.0
	2020-01	-9.1	-12.6	3.5
Pre-intervention	2020-02	-8.1	-9.9	1.8
period	2020-03	-9.5	-14.6	5.1
	2020-04	-12.4	-9.1	-3.3
	2020-05	-21.4	-15.9	-5.5
	2020-06	-24.8	-22.8	-2.0
	2020-07	-22.8	-26.5	3.7
	2020-08	-26.8	-25.6	-1.2
	2020-09	-26.1	-28.4	2.3
	2020-10	-23.4	-31.1	7.7
	2020-11	-26.5	-32.5	6.0
	2020-12	-27.3	-27.1	-0.2
	2021-01	-23.1	-29.3	6.2
	2021-02	-26.5	-29.4	2.9
	2021-03	-21.5	-33.3	11.8
	2021-04	-24.6	-30.0	5.4
	2021-05	-21.4	-23.1	1.7
	2021-06	-21.4	-24.3	2.9
	2021-07	-16.9	-27.8	10.9
D + :	2021-08	-16.4	-26.0	9.6
Post-intervention	2021-09	-17.8	-27.4	9.6
period	2021-10	-14	-29.4	15.4
	2021-11	-18	-31.3	13.3
	2021-12	-19.9	-30.2	10.3
	2022-01	-18.4	-29.1	10.7
	2022-02	-17.3	-39.0	21.7
	2022-03	-25.6	-45.3	19.7
	2022-04	-32.3	-44.9	12.6
	2022-05	-27.6	-45.1	17.5
	2022-06	-33.3	-48.3	15.0
	2022-07	-35.7	-53.0	17.3
	2022-08	-32.1	-47.2	15.1
	2022-09	-31.9	-44.6	12.7
	2022-10	-33.1	-50.4	17.3
	2022-11	-29.3	-49.5	20.2
	2022-12	-29.8	-46.7	16.9

Table 9 – Differences in Subjective Financial Wellbeing Between Spain and Synthetic Spain

# 5.2.3 Statistical Significance of Main Results

To assess the statistical significance of the results, I conduct placebo tests and calculate the post/pre MSPE ratios for each country in the control group to compare it to Spain's ratio. The ratios are displayed in Figure 7. The left-hand panel shows the ratio excluding all placebo cases with a pre-period MSPE exceeding two times the treated unit's pre-period MSPE. The right-

hand panel shows a 'spaghetti plot' displaying the difference in the perceived change in households' financial situation for each country in the donor pool.

The pseudo-P-value is 1/10, meaning there is an 10% chance that the result is achieved at random. The result is statistically significant at conventional levels. I confirm the direction and significance of the results with the RASCM analysis in Section 5.2.4 below.

*Figure 7 – Post/Pre-intervention Mean Squared Prediction Error for Spain and Control Countries* 



#### 5.2.4 Robustness Checks

I perform the same three tests as in Section 5.1.4 to check the robustness of the treatment effect found by the SCM, i.e. that the analysis has been able to correctly reproduce the subjective financial wellbeing trends in Spain in the absence of the IMV.

#### 5.2.4.1 Ridge Augmented Synthetic Control Analysis

To test for the assumption that results are unbiased given an excellent pre-treatment fit, I conduct the same analysis using the RASCM with no covariates. This robustness check is needed since a visual inspection of Figure 6 determines that there are some gaps in the pre-treatment fit of outcome variables between Spain and Synthetic Spain. The analysis corroborates the direction of the SCM results and confirms their statistical significance.

In the RASCM analysis, the pre-treatment fit is improved considerably as seen from the zero difference in the pre-treatment perceived evolution of households' finances between Spain and Synthetic Spain shown in Figure 8. The RASCM confirms the direction of the IMV effect, although this effect is larger in magnitude compared to that found with the SCM. Between June 2020 and December 2022, the IMV increased the balance between those saying their financial situation improved and those saying it deteriorated by 14.6 points. The size of this policy effect over two and a half years corresponds to almost a third of the improvement in the perceived change in financial situation balance that took place during the economic recovery of the financial and debt crises over the six-and-a-half-year period.

The RASCM also shows that this result is statistically significant at conventional levels. There is only a 4.2% chance that the result is found at random. Table 10 presents the weights given to the donor countries used to construct Synthetic Spain with the RASCM.

It is worth noting that the pattern of the IMV effect over time is slightly different compared to that found with the SCM. In the latter, the positive effect of the IMV in mitigating the drop in subjective financial wellbeing of households is sustained in time and keeps increasing while, with the RASCM, this effect seems to start to diminish from February 2022. This different pattern might be a closer reflection of the true IMV effect as the adaptation literature notes the relatively small and short-lived effect of changes in most life circumstances on subjective wellbeing as people adapt relatively fast to their new income situation and increase their aspirations (Brickman, Coates and Janoff-Bulman, 1978; Kahneman and Krueger, 2006; Clark, Frijters and Shields, 2008; Stutzer and Frey, 2010). It would be interesting to monitor if the IMV's improvement to households' perceived financial situation is sustained in time or disappears beyond 2022.

*Figure 8 - Evolution of Differences in Perceived Changes in Financial Situation between Spain and Synthetic Spain* 



*Table 10 - Weights Assigned to Countries Used to Construct Synthetic Spain Using the RASCM* 

	Perceived changes in financial situation
Austria	0.43
Croatia	1.31
Czech Republic	0.28
France	-0.81
Greece	0.77
Hungary	-1.14
Ireland	0.78
Lithuania	-0.40
Netherlands	-0.14
Slovakia	-0.08

#### 5.2.4.2 Other robustness checks

I conduct the 'leave-one-out' test to ensure that results are not driven by the choice of control countries. The exclusion of control units leads to estimates before and after the intervention that are close to those using all donor pool countries, except when Greece is excluded. The magnitude and direction of the IMV effect change as shown in Figure 9. However, the pre-treatment fit also worsens considerably: there is a sixfold increase in the MSPE, which now reaches 50.7. This indicates that the change in results comes not so much from the fact that there are other interventions or large idiosyncratic shocks on the excluded untreated unit, which

would confound my results when including this country, but from the fact that Synthetic Spain cannot accurately track subjective financial wellbeing in real Spain without Greece. Greece and Spain have very similar pre-intervention values of the outcome variable (i.e. an average balance of -14 and -11, respectively) and of the predictor (i.e. an average unemployment rate of 18 and 14, respectively). Excluding the unemployment predictor does not impact results (see Figure 10).

*Figure 9 - Evolution of Subjective Financial Wellbeing for Spain and Synthetic Spain After Excluding Greece from Donor Pool* 



*Figure 10 - Evolution of Subjective Financial Wellbeing for Spain and Synthetic Spain After Excluding Unemployment as a Predictor* 



I perform a time placebo test to check for any changes in behaviour among Spanish people before the IMV was introduced. It could be that households adjust their behaviour after the government announced the IMV in April 2020, meaning its effects could start before its effective implementation in June 2020. I would thus be underestimating the impact of the IMV. I conduct the test by reassigning the treatment status to the months in the pre-intervention period. I also test for the statistical significance of these results. The placebo tests show that the IMV effect on subjective financial wellbeing estimated from June 2020 is larger than the effect found for the months where the treatment was not implemented, which provides significant evidence of an effect of the IMV from June 2020 (Pang, Liu and Xu, 2021). Moreover, none of the placebo interventions yield significant effects.

# **6 DISCUSSION**

Both the SCM and RASCM yearly analyses show that, in 2021, the IMV had no statistically significant effect on objective measures of households' financial wellbeing (i.e. poverty rate, poverty gap, mean income). However, the more nuanced monthly analysis, which accounts for the initial IMV effect from its introduction in June 2020 and allows me to study its effect after 2021, reveals that the IMV helped households feel less pessimistic about the evolution of their finances during the Covid-19 and cost-of-living crises. Between June 2020 and December 2022, the IMV increased the balance between those saying their financial situation improved and those saying it deteriorated by a magnitude of between 10.1 (SCM) and 14.6 (RASCM) points.

While the research approach does not enable me to determine which of the mechanisms presented in Section 2 are behind the results obtained, I offer a few conjectures. In terms of the insignificant results obtained on objective financial wellbeing measures in 2021, it seems that a combination of low levels of adequacy, low take-up, work disincentives and the strategic behaviour of some regions might have been at play. As outlined in Section 2, the IMV benefit levels in 2020 and 2021 were not enough for most household types to surpass the poverty threshold. Additionally, since the policy lacked employment incentive measures and was implemented during a period of limited labour market access, it may have unintentionally disincentivised employment and trapped beneficiaries in poverty. In future research, I assess the work disincentives hypothesis since I study the labour supply effects of the IMV on beneficiaries themselves.

The issue of non-take-up appears to be particularly relevant, especially for explaining the insignificant results on the poverty gap and mean income. Estimates of non-take-up lie at 57%

in Spain for the IMV (Marc *et al.*, 2022). The government (Ministerio de Inclusion, Seguridad Social y Migraciones, 2022b), AIReF (2022) and academics Ayala, Jurado and Perez (2022) have identified several reasons for non-take-up: potential beneficiaries (i) lack information about the existence of the policy; (ii) believe they would not qualify it; (iii) are put off by the complexity of the application process; or (iv) do not find it worth applying given the amount they would receive, especially when they have other support means.

The high non-take-up of the IMV in 2021 might also be due to the slow processing of applications by the administration. As of December 2021, the administration had received 1.5 million submissions, most of them issued in the first three months. The limited administrative capacity meant that during the first few months, the average application processing time was 150 days. Over 40% of applications took more than three months to be resolved. However, the situation has since improved and towards the end of 2021, 80% of applications were now being resolved within 3 months, including a majority being resolved within 1 month (AIReF, 2022).

Another factor explaining the insignificant results obtained on objective financial wellbeing measures that has not been addressed in the literature could be the strategic behaviour of some regions. The introduction of the IMV appears to have incentivised regions to decrease the generosity and coverage of their regional schemes to make savings in their regional budgets. Thus, despite some households receiving the national IMV, current regional MISs beneficiaries might have seen their disposable income decrease, meaning overall national financial wellbeing remains unchanged. According to the State Association of Directors and Managers of Social Services (Asociacion Estatal de Directoras y Gerentes en Servicios Sociales, 2022), evidence suggests that at least five out of seventeen regions, namely Castilla-La-Mancha, Aragón, Extremadura, Madrid and Castilla y León, have reduced the budget allocated to regional MISs by almost 50% in 2021. A further thirteen regions have decreased the amounts of their regional MISs in 2021 compared to 2020 despite the Covid-19 crisis (Ministerio de Derechos Sociales y Agenda 2030, 2022). It is key for the central government to better coordinate the integration of the national IMV with the existent regional MISs.

In terms of the significant IMV effect on subjective financial wellbeing, this might be due to various reasons. On the one hand, as foreseen in Section 2, the provision of a stable source of income through the IMV could have allowed beneficiaries to better plan their expenses, bringing financial stability and reducing stress. The absence of conditionality measures and the

broad public support for the policy may have also contributed to reducing the negative impact of stress and stigma on the subjective financial wellbeing of beneficiaries. Given the positive effect of the IMV, it does not seem to be the case that, as posited in Section 2, beneficiaries have adapted quickly to their new financial situation.

However, it is important to note that these hypotheses assume that individuals are actually receiving the benefit and therefore that the observed effect on subjective financial wellbeing reflects improvements in objective financial wellbeing after 2021. In this sense, the more extensive monthly analysis could be capturing better the lagged rollout of the policy or the effect of the initiatives that the government implemented from 2022 to improve the IMV's take-up, amounts, coverage and to address potential labour supply disincentives. If this is the case, once income data for 2022 becomes available, I should see that the IMV improved objective financial wellbeing indicators.

On the other hand, if this effect of the IMV on objective financial wellbeing after 2021 does not materialise, it would indicate a discrepancy between households' financial status and their perception of their situation. As discussed in Section 2, I anticipated that an improvement in material conditions may not necessarily translate into a perceived improvement in finances given envy among non-recipients and the signalling effect of the IMV's introduction, which may have triggered feelings of economic pessimism among Spanish households. However, the results point to the opposite type of discrepancy, where households perceive that their situation has improved without a corresponding real increase in income.

This mismatch could be explained by the presence of more positive spillover and signalling effects than those found in the literature so far. The government's hasty introduction of the policy during the Covid-19 crisis could have signalled adverse economic conditions to the population, leading them to perceive that they are not faring as poorly as the target population of the IMV. Such findings would contribute to the existing literature on spillover and signalling effects on non-recipients, albeit shedding light on a more empathetic and optimistic view of social relations, challenging previous findings that non-recipients become envious of the perceived gains of others and that emergency policies trigger pessimism among populations.

The discrepancy between households' real financial status and their perception of their situation could also mean that there is an 'anticipation effect' of the policy by which poor households that have already started an application process but whose application has not been processed,

feel their financial situation has improved even though their material conditions remain unchanged. This could fit in a context where there were bottlenecks in the administrative processing of IMV applications. More broadly, the mismatch could also reflect a 'placebo effect' by which households perceive the IMV as a safety net they could access in case of need, thus providing a sense of financial security and reducing uncertainty in a context of economic crisis.

# 7 CONCLUSION

In this paper, I used the case study of Spain and a Synthetic Control Method to examine how MISs affect households' financial wellbeing and whether this effect differs across objective material conditions and households' perceptions. The results show that, while the policy had no statistically significant effect on objective financial wellbeing measures (i.e. the poverty rate, the poverty gap and mean income) in 2021, it did considerably improve subjective financial wellbeing between June 2020 and December 2022, as it helped households feel less pessimistic about the evolution of their finances during the Covid-19 and cost-of-living crises. Between June 2020 and December 2022, the IMV increased the balance between those saying their financial situation improved and those saying it deteriorated by a magnitude of between 10.1 and 14.6 points. Spain's new MIS acted as a lifeline in times of economic uncertainty. Two main questions remain that I will be able test once data becomes available: (i) whether the policy effect on households' perceptions will be sustained over time after December 2022 and (ii) whether it reflects improvements in objective measures beyond 2021.

I have tried to explain the results putting forward several hypotheses that will need to be tested in future research. The insignificant effects of the Spanish minimum income scheme on objective measures in 2021 might be due to the policy's small reach given low levels of takeup, to its low amounts as well as to labour supply reactions of the target population and to the strategic behaviour of regions, which might have offset any positive effect of the policy on national-level financial wellbeing. The relatively large and significant effect of the Spanish MIS on households' perceived financial improvements might reflect the lagged rollout of the policy and the impact of the government's initiatives to improve the policy's adequacy, coverage, take-up and compatibility with work from 2022. These significant results might also reflect disparities between objective and subjective wellbeing, which could be explained by anticipation effects among IMV applicants, placebo effects among potential beneficiaries and/or positive spillover effects among non-recipients. The findings of this paper are relevant to Spanish policymakers as they provide an overview of the achievements and limitations of this new national-level MIS, which has come to be seen as a turning point in the fight against poverty in the country. More generally, the results add empirical causal evidence to the debate on the effects of MISs on financial wellbeing and on the different design elements that make these policies successful. The paper also highlights the importance for practitioners to consider subjective measures of financial wellbeing when assessing income support schemes. Subjective financial wellbeing indicators can reveal important information we might otherwise overlook. In addition, while income data from surveys is available with a two-year lag, some subjective financial wellbeing measures like the one used in this paper can be accessed with only a couple of months' delay, meaning subjective indicators can also be particularly relevant if we are interested in the timely monitoring of minimum income schemes.

# **8 REFERENCES**

- Abadie, A. (2021) 'Using Synthetic Controls: Feasibility, Data Requirements, and Methodological Aspects', *Journal of Economic Literature*, 59(2), pp. 391–425.
- Abadie, A., Diamond, A. and Hainmueller, J. (2010) 'Synthetic Control Methods for Comparative Case Studies: Estimating the Effect of California's Tobacco Control Program', *Journal of the American Statistical Association*, 105(490), pp. 493–505.
- Abadie, A., Diamond, A. and Hainmueller, J. (2015) 'Comparative Politics and the Synthetic Control Method', *American Journal of Political Science*, 59(2), pp. 495–510. Available at: https://doi.org/10.1111/ajps.12116.
- Abadie, A. and Gardeazabal, J. (2003) 'The Economic Costs of Conflict: A Case Study of the Basque Country', *The American Economic Review*, 93(1), pp. 113–132.
- Adhikari, B. (2022) 'A Guide to Using the Synthetic Control Method to Quantify the Effects of Shocks, Policies, and Shocking Policies', *The American Economist*, 67(1), pp. 46–63. Available at: https://doi.org/10.1177/05694345211019714.
- Adhikari, B. and Alm, J. (2016) 'Evaluating the Economic Effects of Flat Tax Reforms Using Synthetic Control Methods', *Southern Economic Journal*, 83(2), pp. 437–463. Available at: https://doi.org/10.1002/soej.12152.
- AIReF (2019) 'Los Programas de Rentas Minimas en Espana: Estudio', AIReF, pp. 1-170.
- AIReF (2022) '1.ª Opinion Ingreso Minimo Vital 1/22'.
- Almeida, V., Poli, S.D. and Hernández, A. (2022) 'The Effectiveness of Minimum Income Schemes in the EU', JRC Working Papers on Taxation & Structural Reforms [Preprint]. Available at: https://ideas.repec.org//p/ipt/taxref/202209.html (Accessed: 8 January 2023).
- Andriopoulou, E. and Tsakloglou, P. (2011) *The Determinants of Poverty Transitions in Europe and the Role of Duration Dependence*. SSRN Scholarly Paper ID 1842089. Rochester, NY: Social Science Research Network. Available at: https://doi.org/10.2139/ssrn.1842089.
- Arriba, A. and Moreno, L. (2005) 'Spain: Poverty, Social Exclusion and "Safety Nets", in Welfare State Reform in Southern Europe: Fighting Poverty and Social Exclusion in Italy, Spain, Portugal and Greece. London: Routledge, pp. 110–162.
- Asociacion Estatal de Directoras y Gerentes en Servicios Sociales (2022) 'Las CC.AA. Redujeron 247 Millones su Gasto en Rentas Mínimas de Inserción en 2021 (-13%), tras la Implantation del IMV', *directoressociales*, 30 December. Available at: https://directoressociales.com/las-cc-aa-redujeron-247-millones-su-gasto-en-rentas-minimas-de-insercion-en-2021-13-tras-la-implantation-del-imv/ (Accessed: 6 January 2023).
- Athey, S. and Imbens, G.W. (2017) 'The State of Applied Econometrics: Causality and Policy Evaluation', *Journal of Economic Perspectives*, 31(2), pp. 3–32. Available at: https://doi.org/10.1257/jep.31.2.3.

- Atsebi, J.-M.B. and Ferrer-i-Carbonell, A. (2022) 'Relative Deprivation in Tanzania: Relative Concerns and Empathy', *Journal of Economic Behavior & Organization*, 198(C), pp. 389– 408.
- Ayala, L. et al. (2021) 'The Effectiveness of Minimum Income Benefits in Poverty Reduction in Spain', International Journal of Social Welfare, 30(2), pp. 152–169. Available at: https://doi.org/10.1111/ijsw.12447.
- Ayala, L., Jurado, A. and Perez, J. (2022) El Ingreso Mínimo Vital: Adecuación y Cobertura, Funcas. Available at: https://www.funcas.es/articulos/el-ingreso-minimo-vital-adecuacion-ycobertura/ (Accessed: 10 December 2022).
- Badenes Plá, N. and Gambau-Suelves, B. (2020) The New 'Minimum Vital Income' in Spain: Distributional and Poverty Effects in the Presence and Absence of Regional Minimum Income Schemes. Working Paper EM 22/20. EUROMOD Working Paper. Available at: https://www.econstor.eu/handle/10419/259511 (Accessed: 12 February 2023).

Baptista, I. et al. (2021) Social Protection and Inclusion Policy Responses to the COVID-19 Crisis. Available at: https://ec.europa.eu/social/main.jsp?langId=en&catId=89&newsId=10065&furtherNews=yes (Accessed: 11 August 2022).

- Bargain, O. and Doorley, K. (2011) 'Caught in the Trap? Welfare's Disincentive and the Labor Supply of Single Men', *Journal of Public Economics*, 95(9), pp. 1096–1110.
- Behrendt, C. (2000) 'Do Means-tested Benefits Alleviate Poverty?: Evidence on Germany, Sweden and the United Kingdom from the Luxembourg Income Study', *Journal of European Social Policy*, 10(1), pp. 23–41. Available at: https://doi.org/10.1177/a011398.
- Ben-Michael, E., Feller, A. and Rothstein, J. (2021) 'The Augmented Synthetic Control Method', Journal of the American Statistical Association, pp. 1–27. Available at: https://doi.org/10.1080/01621459.2021.1929245.
- Bertrand, M. and Mullainathan, S. (2001) 'Do People Mean What They Say? Implications for Subjective Survey Data', *American Economic Review*, 91(2), pp. 67–72. Available at: https://doi.org/10.1257/aer.91.2.67.
- Blank, R.M. et al. (1993) 'Poverty, Income Distribution, and Growth: Are They Still Connected?', Brookings Papers on Economic Activity, 1993(2), pp. 285–339. Available at: https://doi.org/10.2307/2534568.
- Bouvet, F., Bower, R. and Jones, J.C. (2022) 'Currency Devaluation as a Source of Growth in Africa: A Synthetic Control Approach', *Eastern Economic Journal*, 48(3), pp. 367–389. Available at: https://doi.org/10.1057/s41302-022-00211-4.
- Brickman, P., Coates, D. and Janoff-Bulman, R. (1978) 'Lottery Winners and Accident Victims: Is Happiness Relative?', *Journal of Personality and Social Psychology*, 36, pp. 917–927. Available at: https://doi.org/10.1037/0022-3514.36.8.917.

- Brown, S., Taylor, K. and Wheatley Price, S. (2005) 'Debt and Distress: Evaluating the Psychological Cost of Credit', *Journal of Economic Psychology*, 26(5), pp. 642–663.
- Brunori, P., Chiuri, M. and Peragine, V. (2010) *The Economic Effects of a Local Minimum Income Support Program.* SERIES 0029. Dipartimento di Economia e Finanza Università degli Studi di Bari 'Aldo Moro'. Available at: https://econpapers.repec.org/paper/baiseries/economia-series29.htm (Accessed: 24 March 2022).
- Bryman, A. (2016) Social Research Methods. 5th edn. Oxford: Oxford University Press.
- Canova, L., Piccoli, L. and Spadaro, A. (2015) 'An Ex-ante Evaluation of the Revenu de Solidarité Active by Micro-macro Simulation Techniques', *IZA Journal of European Labor Studies*, 4(1), p. 17. Available at: https://doi.org/10.1186/s40174-015-0040-3.
- Cantillon, B., Goedemé, T. and Hills, J. (eds) (2019) Decent Incomes for All: Improving Policies in Europe. 1st edn. Oxford University Press. Available at: https://doi.org/10.1093/oso/9780190849696.001.0001.
- Charotti, C.J., Palma, N. and Santos, J.P. dos (2022) American Treasure and the Decline of Spain, Economics Discussion Paper Series. 2201. Economics, The University of Manchester. Available at: https://ideas.repec.org/p/man/sespap/2201.html (Accessed: 16 September 2022).
- CIS (2020) 'Barometro de Mayo 2020', Centro de Investigaciones Sociologicas, #3281, p. 26.
- Clark, A.E., Frijters, P. and Shields, M.A. (2008) 'Relative Income, Happiness, and Utility: An Explanation for the Easterlin Paradox and Other Puzzles', *Journal of Economic Literature*, 46(1), pp. 95–144.
- Council of the European Union (2023) *Council Adopts Recommendation on Adequate Minimum Income*. Available at: https://www.consilium.europa.eu/en/press/pressreleases/2023/01/30/council-adopts-recommendation-on-adequate-minimum-income/ (Accessed: 13 June 2023).
- Currie, J. (2004) *The Take Up of Social Benefits*. Working Paper 10488. National Bureau of Economic Research. Available at: https://doi.org/10.3386/w10488.
- D'Acunto, F., Hoang, D. and Weber, M. (2019) 'Managing Households' Expectations with Simple Economic Policies'. Rochester, NY. Available at: https://doi.org/10.2139/ssrn.3439442.
- Danziger, S. and Gottschalk, P. (1995) America Unequal. New York: Russell Sage Foundation.
- Davis, O. (2019) 'What is the Relationship between Benefit Conditionality and Mental Health? Evidence from the United States on TANF Policies', *Journal of Social Policy*, 48(2), pp. 249–269. Available at: https://doi.org/10.1017/S0047279418000363.
- Di Tella, R. and MacCulloch, R. (2006) 'Some Uses of Happiness Data in Economics', *Journal of Economic Perspectives*, 20(1), pp. 25–46. Available at: https://doi.org/10.1257/089533006776526111.

- Dolan, P., Peasgood, T. and White, M. (2008) 'Do We Really Know What Makes Us Happy? A Review of the Economic Literature on the Factors Associated with Subjective Well-being', *Journal of Economic Psychology*, 29(1), pp. 94–122.
- Dustmann, C., Frattini, T. and Halls, C. (2010) 'Assessing the Fiscal Costs and Benefits of A8 Migration to the UK', *Fiscal Studies*, 31(1), pp. 1–41.
- EAPN (2021) El Ingreso Mínimo Vital un Año Después. La Perspectiva Autonómica, EAPN España. Available at: https://www.eapn.es/publicaciones/452/el-ingreso-minimo-vital-un-anodespues-la-perspectiva-autonomica (Accessed: 12 February 2023).
- Easterlin, R.A. (1974) *Does Economic Growth Improve the Human Lot? Some Empirical Evidence*. Available at: https://mpra.ub.uni-muenchen.de/111773/ (Accessed: 7 January 2023).
- Esaka, T. and Fujii, T. (2022) 'Quantifying the Impact of the Tokyo Olympics on COVID-19 Cases using Synthetic Control Methods', *Journal of the Japanese and International Economies*, 66, p. 101228. Available at: https://doi.org/10.1016/j.jjie.2022.101228.
- European Commission (2019) 2019 European Semester: Country Specific Recommendations / Commission Recommendations, European Commission. Available at: https://ec.europa.eu/info/publications/2019-european-semester-country-specificrecommendations-commission-recommendations\_en (Accessed: 25 August 2022).
- European Commission (2022a) Fairness, Inequality and Inter-Generational Mobility, Eurobarometer. Available at: https://europa.eu/eurobarometer/surveys/detail/2652 (Accessed: 19 February 2023).
- European Commission (2022b) *Minimum Income: More Effective Support Needed to Fight Poverty and Promote Employment*. Available at: https://ec.europa.eu/social/main.jsp?langId=en&catId=89&furtherNews=yes&newsId=10417 #navItem-relatedDocuments (Accessed: 9 October 2022).
- European Commission, ICF GHK and Milieu Ltd (2013) 'A Fact Finding Analysis on the Impact on the Member States' Social Security Systems of the Entitlements of Non-Active Intra-EU migrants to Special Non-Contributory Cash Benefits and Healthcare Granted on the Basis of Residence: Final Report.', *Publications Office of the European Union*, pp. 1–282.
- Ferman, B. and Pinto, C. (2021) Synthetic Controls with Imperfect Pre-Treatment Fit, Papers. 1911.08521. arXiv.org. Available at: https://ideas.repec.org/p/arx/papers/1911.08521.html (Accessed: 22 September 2022).
- Figari, F., Matsaganis, M. and Sutherland, H. (2013) 'Are European Social Safety Nets Tight Enough? Coverage and Adequacy of Minimum Income Schemes in 14 EU Countries', *International Journal of Social Welfare*, 22(1), pp. 3–14. Available at: https://doi.org/10.1111/j.1468-2397.2012.00885.x.
- Firpo, S. and Possebom, V. (2018) 'Synthetic Control Method: Inference, Sensitivity Analysis and Confidence Sets', *Journal of Causal Inference*, 6(2). Available at: https://econpapers.repec.org/article/bpjcausin/v\_3a6\_3ay\_3a2018\_3ai\_3a2\_3ap\_3a26\_3an\_3 a1.htm (Accessed: 8 September 2022).

- Fleche, S. *et al.* (2012) 'Exploring Determinants of Subjective Wellbeing in OECD Countries: Evidence from the World Values Survey', *Journal of Poverty and Social Justice*, 20(2), pp. 227–233. Available at: https://doi.org/10.1332/175982712X652122.
- Frazer, H. and Marlier, E. (2016) *Minimum Income Schemes in Europe: A Study of National Policies* 2015. European Commission, p. 63.
- Frijters, P. and Krekel, C. (2021) 'Wellbeing Measurement and Policy Design—Measures, Key Findings, and Wellbeing Frameworks', in P. Frijters and C. Krekel (eds) A Handbook for Wellbeing Policy-Making: History, Theory, Measurement, Implementation, and Examples. Oxford University Press, p. 0. Available at: https://doi.org/10.1093/oso/9780192896803.003.0002.
- Gallo, G. (2021) 'Regional Support for the National Government: Joint Effects of Minimum Income Schemes in Italy', *Italian Economic Journal*, 7(1), pp. 149–185. Available at: https://doi.org/10.1007/s40797-019-00118-8.
- Gilchrist, D. *et al.* (2022) 'Synthetic Control Method: A tool for Comparative Case Studies in Economic History', *Journal of Economic Surveys*, n/a(n/a), p. 37. Available at: https://doi.org/10.1111/joes.12493.
- Goldfayn-Frank, O., Kocharkov, G. and Weber, M. (2020) Households' Expectations and Unintended Consequences of Policy Announcements, Bundesbank. Available at: https://www.bundesbank.de/en/publications/research/research-brief/2020-34-householdsexpectations-849832 (Accessed: 5 January 2023).
- Gorjón, L. and Villar, A. (2019) 'The Minimum Income Scheme as a Poverty Reduction Mechanism: The Case of the Basque Country', *ISEAK*, p. 22.
- Grigoli, F., Koczan, Z. and Topalova, P. (2018) 'Drivers of Labor Force Participation in Advanced Economies: Macro and Micro Evidence', *IMF Working Papers*, 18(150), p. 1. Available at: https://doi.org/10.5089/9781484361528.001.
- Hale, T. *et al.* (2021) 'A Global Panel Database of Pandemic Policies (Oxford COVID-19 Government Response Tracker)', *Nature Human Behaviour*, 5(4), pp. 529–538. Available at: https://doi.org/10.1038/s41562-021-01079-8.
- Haushofer, J. and Shapiro, J. (2016) 'The Short-term Impact of Unconditional Cash Transfers to the Poor: Experimental Evidence from Kenya', *The Quarterly Journal of Economics*, 131(4), pp. 1973–2042. Available at: https://doi.org/10.1093/qje/qjw025.
- Hoynes, H., Page, M. and Stevens, A. (2005) Poverty in America: Trends and Explanations. Working Paper 11681. National Bureau of Economic Research. Available at: https://doi.org/10.3386/w11681.
- ILO (2022) Country Policy Responses (COVID-19 and the World of Work). Available at: https://www.ilo.org/global/topics/coronavirus/regional-country/country-responses/lang-en/index.htm (Accessed: 11 August 2022).

- Iramani, R. and Lutfi, L. (2021) 'An Integrated Model of Financial Well-Being: The Role of Financial Behavior', *Accounting*, 7(3), pp. 691–700.
- Jacobsen, B., Lee, J.B. and Marquering, W. (2008) 'Are Men More Optimistic?' Rochester, NY. Available at: https://papers.ssrn.com/abstract=1030478 (Accessed: 14 April 2023).
- Jefferis, B.J. et al. (2011) 'Associations Between Unemployment and Major Depressive Disorder: Evidence from an International, Prospective Study (the Predict Cohort)', Social Science & Medicine (1982), 73(11), pp. 1627–1634. Available at: https://doi.org/10.1016/j.socscimed.2011.09.029.
- Jenkins, S., Sacker, A. and Taylor, M. (2011) 'Financial Capability, Income and Psychological Wellbeing', *ISER Working Paper Series* [Preprint]. Available at: https://ideas.repec.org//p/ese/iserwp/2011-18.html (Accessed: 7 January 2023).
- Joo, S. and Grable, J.E. (2004) 'An Exploratory Framework of the Determinants of Financial Satisfaction', *Journal of Family and Economic Issues*, 25(1), pp. 25–50. Available at: https://doi.org/10.1023/B:JEEI.0000016722.37994.9f.
- Jun, M. (2019) 'Stigma and Shame Attached to Claiming Social Assistance Benefits: Understanding the Detrimental Impact on UK Lone Mothers' Social Relationships', *Journal of Family Studies*, 28, pp. 1–17. Available at: https://doi.org/10.1080/13229400.2019.1689840.
- Kahneman, D. and Krueger, A.B. (2006) 'Developments in the Measurement of Subjective Well-Being', *Journal of Economic Perspectives*, 20(1), pp. 3–24. Available at: https://doi.org/10.1257/089533006776526030.
- Kingdon, G.G. and Knight, J. (2007) 'Community, Comparisons and Subjective Well-Being in a Divided Society', *Journal of Economic Behavior & Organization*, 64(1), pp. 69–90. Available at: https://doi.org/10.1016/j.jebo.2007.03.004.
- Kleven, H.J. and Kopczuk, W. (2011) 'Transfer Program Complexity and the Take-Up of Social Benefits', *American Economic Journal: Economic Policy*, 3(1), pp. 54–90. Available at: https://doi.org/10.1257/pol.3.1.54.
- Kuhn, P. et al. (2011) 'The Effects of Lottery Prizes on Winners and Their Neighbors: Evidence from the Dutch Postcode Lottery', American Economic Review, 101(5), pp. 2226–2247. Available at: https://doi.org/10.1257/aer.101.5.2226.
- La Moncloa (2021) *El Ingreso Mínimo Vital Llega en Septiembre a 800.000 Personas*. Available at: https://www.lamoncloa.gob.es/serviciosdeprensa/notasprensa/inclusion/Paginas/2021/011021 -imv-beneficiarios.aspx (Accessed: 15 March 2022).
- Layard, R. and De Neve, J.-E. (2023) *Wellbeing: Science and Policy*. Cambridge: Cambridge University Press. Available at: https://doi.org/10.1017/9781009298957.
- Lemieux, T. and Milligan, K. (2008) 'Incentive Effects of Social Assistance: A Regression Discontinuity Approach', *Journal of Econometrics*, 142(2), pp. 807–828.

- Lewis, D.J., Makridis, C. and Mertens, K. (2019) 'Do Monetary Policy Announcements Shift Household Expectations?' Rochester, NY. Available at: https://doi.org/10.2139/ssrn.3449467.
- Lister, R. (2004) Poverty. Cambridge: Polity Press.
- Lundberg, O. *et al.* (2010) 'The Potential Power of Social Policy Programmes: Income Redistribution, Economic Resources and Health', *International Journal of Social Welfare*, 19(s1), pp. S2–S13. Available at: https://doi.org/10.1111/j.1468-2397.2010.00727.x.
- Lundberg, O. (2016) 'Social Protection, Income and Health Inequities', in. Available at: https://www.semanticscholar.org/paper/Social-protection-%2C-income-and-health-inequities-Lundberg/baf966734dc44b234da2858047623d61ad8b4894 (Accessed: 13 February 2023).
- Marc, C. *et al.* (2022) 'Quantifier le Non-recours aux Minima Sociaux en Europe', *Direction de la Recherche, des etudes, de l'evaluation et des statistiques* [Preprint]. Available at: https://drees.solidarites-sante.gouv.fr/publications-communique-de-presse/les-dossiers-de-la-drees/quantifier-le-non-recours-aux-minima (Accessed: 5 January 2023).
- McClelland, R. and Mucciolo, L. (2022) 'An Update on the Synthetic Control Method as a Tool to Understand State Policy', *Urban Institute* [Preprint]. Available at: https://www.urban.org/research/publication/update-synthetic-control-method-toolunderstand-state-policy (Accessed: 8 September 2022).
- McGinty, E.E. *et al.* (2022) 'Effects of State Opioid Prescribing Laws on Use of Opioid and Other Pain Treatments Among Commercially Insured U.S. Adults', *Annals of Internal Medicine*, 175(5), pp. 617–627. Available at: https://doi.org/10.7326/M21-4363.
- Melosi, L., Morita, H. and Zanetti, F. (2022) 'The Signaling Effects of Fiscal Announcements'. Rochester, NY. Available at: https://doi.org/10.2139/ssrn.4205208.
- Milligan, K. and Stabile, M. (2011) 'Do Child Tax Benefits Affect the Well-Being of Children? Evidence from Canadian Child Benefit Expansions', *American Economic Journal: Economic Policy*, 3(3), pp. 175–205. Available at: https://doi.org/10.1257/pol.3.3.175.
- Ministerio de Derechos Sociales y Agenda 2030 (2022) *Informe de Rentas Mínimas de Inserción*. Available at: https://www.mdsocialesa2030.gob.es/derechos-sociales/serviciossociales/rentas-minimas.htm (Accessed: 18 March 2022).
- Ministerio de Inclusion, Seguridad Social y Migraciones (2022a) *Revista de la Seguridad Social*, *Revista Seguridad Social*. Available at: https://revista.seg-social.es/-/20221229-imv (Accessed: 5 January 2023).
- Ministerio de Inclusion, Seguridad Social y Migraciones (2022b) Un Dia en el Autobus del IMV, La Revista de la Seguridad Social. Available at: https://revista.seg-social.es/-/20221028-un-dia-en-bus-imv (Accessed: 6 January 2023).
- MISSOC (2022) 'Comparative Tables'. Available at: https://www.missoc.org/missocdatabase/comparative-tables/ (Accessed: 7 September 2022).

- Moffitt, R. (1983) 'An Economic Model of Welfare Stigma', *American Economic Review*, 73(5), pp. 1023–35.
- Moffitt, R.A. (2016) 'The US Safety Net and Work Incentives: Is There a Problem? What Should be Done?', *The US Labor Market: Questions and Challenges for Public Policy* [Preprint].
- Muir, K. *et al.* (2017) 'Exploring Financial Wellbeing in the Australian Context', *Financial Literacy Australia* [Preprint].
- Murray, C. (1984) Losing Ground: American Social Policy, 1930-1980. New York: Basic Books.
- Notten, G. and Guio, A.-C. (2016) The Impact of Social Transfers on Income Poverty and Material Deprivation. ImPRovE Working Paper 16/17. Herman Deleeck Centre for Social Policy, University of Antwerp. Available at: https://econpapers.repec.org/paper/hdlimprov/1617.htm (Accessed: 23 December 2022).
- OECD (2010) Making Reform Happen Lessons from OECD Countries: Lessons from OECD Countries. OECD Publishing.
- OECD (2020) Supporting Livelihoods During the COVID-19 Crisis: Closing the Gaps in Safety Nets, OECD. Available at: https://www.oecd.org/coronavirus/policy-responses/supportinglivelihoods-during-the-covid-19-crisis-closing-the-gaps-in-safety-nets-17cbb92d/ (Accessed: 11 August 2022).
- Pang, X., Liu, L. and Xu, Y. (2021) A Bayesian Alternative to Synthetic Control for Comparative Case Studies. SSRN Scholarly Paper ID 3649226. Rochester, NY: Social Science Research Network. Available at: https://doi.org/10.2139/ssrn.3649226.
- Peña-Casas, R. and Bouget, D. (2013) 'Towards a European Minimum Income? Discussions, Issues and Prospects', *European Economic and Social Committee*, No EESC/COMM/03/2013, pp. 85–104.
- Peri, G. and Yasenov, V. (2015) The Labor Market Effects of a Refugee Wave: Applying the Synthetic Control Method to the Mariel Boatlift. Working Paper 21801. National Bureau of Economic Research. Available at: https://doi.org/10.3386/w21801.
- Portney, P. and Mead, L. (1990) 'Should Workfare be Mandatory? What Research Says', *Journal of Policy Analysis and Management*, 9(3), pp. 400–404. Available at: https://doi.org/10.2307/3325284.
- Rodrigues, C.F. (2004) *The Redistributive Impact of the Guaranteed Minimum Income Programme in Portugal, Working Papers Department of Economics.* 2004/09. ISEG - Lisbon School of Economics and Management, Department of Economics, Universidade de Lisboa. Available at: https://ideas.repec.org/p/ise/isegwp/wp92004.html (Accessed: 24 March 2022).
- Salignac, F. *et al.* (2020) 'Conceptualizing Financial Wellbeing: An Ecological Life-Course Approach', *Journal of Happiness Studies*, 21(5), pp. 1581–1602. Available at: https://doi.org/10.1007/s10902-019-00145-3.

- Seghieri, C., Desantis, G. and Tanturri, M. (2006) 'The Richer, the Happier? An Empirical Investigation in Selected European Countries', *Social Indicators Research: An International and Interdisciplinary Journal for Quality-of-Life Measurement*, 79(3), pp. 455–476.
- Simpson, J. et al. (2021) 'Effects of Social Security Policy Reforms on Mental Health and Inequalities: A Systematic Review of Observational Studies in High-Income Countries', Social Science & Medicine [Preprint]. Available at: https://doi.org/10.1016/j.socscimed.2021.113717.
- Stutzer, A. and Frey, B.S. (2010) *Happiness and Economics: How the Economy and Institutions Affect Human Well-Being*. Princeton: Princeton University Press. Available at: https://muse.jhu.edu/pub/267/monograph/book/30388 (Accessed: 7 January 2023).
- Sutherland, H. (2017) 'Quality Assessment of Microsimulation Models: The Case of EUROMOD', *International Journal of Microsimulation*, 11(1), pp. 198–223. Available at: https://doi.org/10.34196/ijm.00178.
- Tasseva, I.V. (2016) 'Evaluating the Performance of Means-Tested Benefits in Bulgaria', *Journal of Comparative Economics*, 44(4), pp. 919–935. Available at: https://doi.org/10.1016/j.jce.2016.02.003.
- Thom, M. (2022) 'Can Additional Funding Improve Mental Health Outcomes? Evidence from a Synthetic Control Analysis of California's Millionaire Tax', *PLOS ONE*, 17(7), p. e0271063. Available at: https://doi.org/10.1371/journal.pone.0271063.
- Thornton, I. and Iacoella, F. (2022) 'Conditionality and Contentment: Universal Credit and UK Welfare Benefit Recipients' Life Satisfaction', *Journal of Social Policy*, pp. 1–29. Available at: https://doi.org/10.1017/S0047279422000241.
- Tomasi, S. (2022) 'The (Non) Impact of the Spanish "Tax on the Sun" on Photovoltaics Prosumers Uptake', *Energy Policy*, 168, p. 113041. Available at: https://doi.org/10.1016/j.enpol.2022.113041.
- West, T., Cull, M. and Johnson, D. (2021) 'Income More Important Than Financial Literacy for Improving Wellbeing', *Financial Services Review*, pp. 187–207.
- Wickham, S. *et al.* (2020) 'Effects on Mental Health of a UK Welfare Reform, Universal Credit: A Longitudinal Controlled Study', *The Lancet Public Health*, 5(3), pp. e157–e164. Available at: https://doi.org/10.1016/S2468-2667(20)30026-8.
- Winter, M. et al. (1999) 'Constraints, Domain Conditions, and Well-Being: Evidence from Poland during the Transformation', *The Journal of Consumer Affairs*, 33(1), pp. 27–47.
- Xiao, J.J. (2013) 'Family Economic Well-Being', in G.W. Peterson and K.R. Bush (eds) Handbook of Marriage and the Family. Boston, MA: Springer US, pp. 573–611. Available at: https://doi.org/10.1007/978-1-4614-3987-5\_24.

# 9 APPENDIX

# 9.1 EU Countries with Policies Similar to the Spanish IMV

Country	Measure	Timing	Target Population
Belgium	Extra allowance to social assistance	Jul 2020 – Jun 2021	Beneficiaries of social
	beneficiaries		assistance including the
			minimum income benefit
	Annual raise of social security and social	Jan 2021 – Present	Beneficiaries of social security
	assistance benefits in the direction of the at-		and assistance including the
	risk-of-poverty threshold		minimum income benefit
Finland	Top-up in social assistance benefits	Sept – Dec 2020	Beneficiaries of basic social
			assistance
Italy	Introduction of an emergency income (RdE)	May 2020 - Present	Poor households not covered by
	that can be received for a maximum of 5		other ordinary or extraordinary
	months		benefits
Latvia	Increase in guaranteed minimum income	Jan – Dec 2021	Beneficiaries of guaranteed
			minimum income
Luxembourg	Doubling of the high-cost-of-living	Jan – Dec 2020	Beneficiaries of high-cost- of-
	allowance		living allowance
Portugal	Change in reference period for the	Jul – Dec 2020	Recipients and potential
	calculation of the social insertion income		recipients of social insertion
			income

Source: Own construction from Baptista et al. (2021), OECD (2020), ILO (2022) and MISSOC (2022)

# 9.2 'LEAVE-ONE-OUT' TESTS WITH ESTIMATE CHANGES & CORRESPONDING (IN)SIGNIFICANCE OF RESULTS



Exclusion of Romania





# 9.3 IN-TIME PLACEBO TESTS WITH POSITIVE RESULTS & CORRESPONDING (IN)SIGNIFICANCE OF RESULTS

63







